Mississippi State University Scholars Junction

Theses and Dissertations

Theses and Dissertations

12-10-2005

Empirical Findings On Persuasiveness Of Recommender Systems For Customer Decision Support In Electronic Commerce

Qinyu Liao

Follow this and additional works at: https://scholarsjunction.msstate.edu/td

Recommended Citation

Liao, Qinyu, "Empirical Findings On Persuasiveness Of Recommender Systems For Customer Decision Support In Electronic Commerce" (2005). *Theses and Dissertations*. 1865. https://scholarsjunction.msstate.edu/td/1865

This Dissertation - Open Access is brought to you for free and open access by the Theses and Dissertations at Scholars Junction. It has been accepted for inclusion in Theses and Dissertations by an authorized administrator of Scholars Junction. For more information, please contact scholcomm@msstate.libanswers.com.

EMPIRICAL FINDINGS ON PERSUASIVENESS OF RECOMMENDER SYSTEMS FOR CUSTOMER DECISION SUPPORT IN ELECTRONIC COMMERCE

By

Qinyu Liao

A Dissertation Submitted to the Faculty of Mississippi State University in Partial Fulfillment of the Requirements for the Degree of Doctor of Philosophy in Business Administration in the Department of Management and Information Systems

Mississippi State University

December 2005

Copyright by

Qinyu Liao

2005

EMPIRICAL FINDINGS ON PERSUASIVENESS OF RECOMMENDER SYSTEMS FOR CUSTOMER DECISION SUPPORT IN ELECTRONIC

COMMERCE

By

Qinyu Liao

Approved:

Jung P. Shim Professor of Management and Information Systems (Director of Dissertation) Merrill Warkentin Professor of Management and Information Systems (Committee Member)

Kirk P. Arnett Professor of Management and Information Systems (Committee Member) Ronald Taylor Professor of Marketing/ Quantitative Analysis/Business Law (Committee Member)

Gary Templeton Assistant Professor of Management and Information Systems (Committee Member)

Barbara A. Spencer Professor of Management and Information Systems Director of Graduate Studies in the College of Business and Industry

Sara M. Freedman Dean of College of Business and Industry Name: Qinyu Liao

Date of Degree: December 9, 2005

Institution: Mississippi State University

Major Field: Business Administration in Management Information Systems

Major Professor: Dr. Jung P. Shim

Title of Study: EMPIRICAL FINDINGS ON PERSUASIVENESS OF RECOMMENDER SYSTEMS FOR CUSTOMER DECISION SUPPORT IN ELECTRONIC COMMERCE

Pages in Study: 151

Candidate for Degree of Doctor of Philosophy

More and more companies are making online presence by opening online stores and providing customers with company and products information but the overwhelming amount of information also creates information overload for the customers. Customers feel frustrated when given too many choices while companies face the problem of turning browsers into actual buyers. Online recommender systems have been adopted to facilitate customer product search and provide personalized recommendation in the market place. The study will compare the persuasiveness of different online recommender systems and the factors influencing customer preferences.

Review of the literature does show that online recommender systems provide customers with more choices, less effort, and better accuracy. Recommender systems using different technologies have been compared for their accuracy and effectiveness. Studies have also compared online recommender systems with human recommendations and recommendations from expert systems. The focus of the comparison in this study is on the recommender systems using different methods to solicit product preference and develop recommendation message. Different from the technology adoption and acceptance models, the persuasive theory used in the study is a new perspective to look at the end user issues in information systems.

This study will also evaluate the impact of product complexity and product involvement on recommendation persuasiveness. The goal of the research is to explore whether there are differences in the persuasiveness of recommendation given by different recommender systems as well as the underlying reasons for the differences. Results of this research may help online store designers and ecommerce participants in selecting online recommender systems so as to improve their products target and advertisement efficiency and effectiveness.

DEDICATION

I would like to dedicate this research to my parents, who have been the source of strength and support through out this doctoral program. It would have been impossible to follow this dream of becoming a college professor if it were not for their constant and loving support.

ACKNOWLEGEMENTS

I would like to express my heartfelt gratitude to the members of my committee for their support and assistance: my dissertation chair, Dr. J. P. Shim, Dr. Merrill Warkentin, Dr. Kirk Arnett, Dr. Ronald Taylor, and Dr. Gary Templeton. Mississippi State University has great faculty, and I am proud to say that I had some of the finest on my committee. They taught me that dedication and perseverance are important attributes in becoming a respected academician. Too numerous to mention are the other faculty members who shared knowledge with me. A special word of thanks goes to Dr. Steve Taylor and Dr. J. Martin Giesen for the suggestions they provided me during the dissertation process.

I would also like to thank Lin Zhang, Xin Luo, Xin Bu, Linnea Hall, Dr. Jixiang Wu for their friendship during the doctoral program and dissertation process. I am glad that I had close friends like Xin and Lin every step of the program to share my joys and console me in my failures.

TABLE OF CONTENTS

DEDICATION	ii
ACKNOWLEDGEMENTS	iii
LIST OF TALBES	vi
LIST OF FIGURES	viii
CHAPTER	
I. INTRODUCTION	1
Overview	1
Evolution of Internet-Based Electronic Commerce	7
Development of Decision Support Systems and Intelligent Agents	9
Online Recommender Systems	12
Persuasion Theories	18
Research Questions	22
Research Methodology	23
Research Sample and Design	23
Covariate	24
Research Instrument	25
Data Analysis	27
Implications of the Research	28
Organization of the Study	30
II. RECOMMENDER SYSTEMS AND HYPOTHESES DEVELOPMENT	32
Recommender Systems in E-commerce Characteristics of Recommender Systems Approaches Used in Recommender Systems	32 35 36 41
Persuasion and Acceptance-Consistency-Effectiveness (ACE) Model	44

CHAPTER

Page

	Product Complexity Research Hypotheses	52 54
III.	RESEARCH METHODOLOGY	59
	Pilot Study Research Subjects	59 60 61 62 65 68 69 75
	Data Analysis and Statistical Procedure	77
IV.	DATA ANALYSIS	81
	Sample Characteristics Data Analysis Scree Plot and Exploratory Factor Analysis Internal Consistency MANCOVA	82 84 84 87 89
	Evaluation of the Research Hypotheses	95
V.	CONCLUSION AND IMPLICATIONS	101 108
BIBLI	OGRAPHY	112
APPEN	NDIX	
A. REC	COMMENDER SYSTEM EXPERIMENT SURVEY	132

LIST OF TABLES

TABL	ιE	Page
1.	Summary of Recommender System Research	5
2.	Comparison of Filtering Algorithms Used by Recommender Systems	44
3.	Operationalization of Research Variables	75
4.	Sample Demographics	83
5.	Exploratory Factor Analysis	85
6.	Factor Analysis After Items Elimination	87
7.	Cronbach's Coefficient Alpha Values for Research Constructs	88
8.	MANCOVA of Research Model	90
9.	MANCOVA of Research Model (without product complexity)	91
10.	Comparison of Means for Recommender Systems	92
11.	Comparison of Means for Recommender System & Product Complexity	93
12.	ANCOVA of Persuasiveness	94
13.	Comparison of Mean for H _{1a}	96
14.	Comparison of Means for H _{1b}	96
15.	Comparison of Means for H _{1c}	96
16.	Comparison of Means for H ₂	97
17.	Comparison of Means for H _{2a}	97
18.	Comparison of Means for H _{2b}	97

TABL	LE	Page
19.	Comparison of Means for H _{2c}	98
20.	Summary of Hypothesis Testing	100

LIST OF FIGURES

FIGURE	Page
1. New Customer Recommender Interface at Amazon.com	16
2. Customer with Rated Items for Recommendation Building	17
3. Customer Product Attributes Constraints Input at Dell.com	18
4. Research Model	55

CHAPTER I

INTRODUCTION

The purpose of this chapter is to lay the foundation for a dissertation. This chapter provides an overview of electronic commerce, decision support systems and intelligent agents, online recommendation systems, and the persuasion theories. The research problem statement, research questions, and research methodology are included. Finally, the implications of the study and organization of the dissertation are presented.

Overview

With the growing popularity of the Internet and the prosperity of electronic commerce (e-commerce), traditional trading behavior patterns have significantly changed (GVU, 1999). For instance, pre-purchase information searching and online shopping are becoming more popular. However, the exponentially increasing information provided by Internet enterprises is both a blessing and a curse. More information may allow customers to select product options that better match their personal preferences than they would otherwise. On the other hand, having access to too much information may cause information overload and frustration with the different information sources provided by the Internet. While online businesses compete for customer time and attention, information overload can be a threat to customer satisfaction and loyalty (Lee, Liu, and Lu, 2002; Maes, Butmann, and Moukas, 1999; Reibstein, 2002). Online businesses need to provide exceptional value and services based on customer needs if they want customer information browsing at their online store to finally lead to persuaded purchase behavior. One way to overcome such a problem is to build personalized recommender systems using intelligent agents and good user interfaces to retrieve product information that really interests the customers (Desharnais, Lu, and Radhakrishnan, 2002).

Recommender systems are interactive decision support systems that assist consumers in the initial screening of alternatives available in online stores (Haubl, and Trifts, 2000). A highly persuasive recommender system will work as a persuasive salesperson. They provide recommendations for product search and selection (Detlor, and Arsenault, 2002; O'Keefe, and McEachern, 1998), product customization (Grenci, and Todd, 2002) and tell the customer what to buy or who to buy from (Ansari, Essegaier, and Kohli, 2000; Maes *et al.*, 1999). The products can be recommended based on the top overall sellers on a site, on the demographics of the consumer, or on an analysis of the past buying behavior of the consumer as a prediction for future buying behavior. Recommender systems enable customers to cope with information complexity and information overload (Chiasson, and Dexter, 2001; Hanani, Shaira, and Shoval, 2001; Nwana, and Azarmi, 1997), reduce effort, improve the effectiveness and efficiency of customer decision-making (Haubl *et al.*, 2000; Lynch, and Ariely, 2000), and help to configure customized products (Grenci *et al.*, 2002). Different forms of recommender systems are currently implemented on the website of a number of online retail stores (e.g., www.macys.com, www.netmarket.com, www.amazon.com, www.dell.com, www.personalogic.com).

One of the central issues regarding recommender systems is their *persuasiveness* (Komiak, and Benbasat, 2004). Persuasiveness refers to the extent to which customers are moved or influenced by a given recommender system, which is defined here as the reasoning provided by recommender systems regarding the fit of product features with personal needs. Persuasiveness of decision support systems/recommender systems may vary when customer expertise and product complexity vary (Gregor, and Benbasat, 1999; Huang, Chung, Barbara, and Chen, 2004; Jiang, Klein, and Vedder, 2000; Mao, and Benbasat, 2000). Persuasiveness reflects the power of a recommender system as a customer decision support system. It is important because customer (particularly those with limited expertise) confidence in the recommender system, self-confidence in decisions, and the differences between customer and recommender system opinions are all related to the persuasive power of the system (Jiang et al., 2000). Decision support systems are usually not used efficiently by decision makers because of a lack of confidence in the recommendations they provide (Moulin, Irandoust, Belanger, and Desbordes, 2002). In the context of e-commerce customer purchase decision making, a time lag usually exists between the time such a decision must be made and when the customer can physically obtain the product and become aware of its quality. As a result, the acceptance of a recommender system product recommendation is more likely to be

determined by its persuasiveness rather than by its correctness (The match between the product delivered and their searching criteria). Therefore, it is important for a recommender system to convince the customer that its recommendation is relevant, justified and useful. If designers and owners of online stores know which type of recommender system is more persuasive for certain type of products and why, more persuasive recommender systems could be designed and built.

Over the past decade, a number of researchers have studied the influence of trust, explanation and transparency on end user confidence in decisions provided by recommender systems as well as other types of decision support systems. Table 1 showed a list of empirical researches done with research method and results on the use of recommender systems (Sinha, and Swearingen, 2002; Wang, and Benbasat, 2003; Ye, and Johnson, 1995)

· ·	Table 1	Summary	of Reco	ommender	Systems	Research
-	1 4010 1.	Sammary	01 1000	ommender	Sjocenno	researen

Study	Research	Results
	method	
Todd and	Lab	The use of a decision aid may result in effort saving but not improved decision
Benbasat, 1992	experiment	performance.
Morris, 1994	Lab	An expert system-based tool is preferred for company information databases search in
	experiment	terms of accuracy and consistency of recommendations, ease of use, confidence in
	-	recommendation and time taken.
Ye and	Lab	Explanation of reasoning process convinces decision maker of the recommendation
Johnson,1995	experiment	soundness.
O'Keefe and	Case study	Development of an online customer decision-making process to suggest points of
McEachern,		recommendation agent support.
1998		
Schafer et al.,	Case study	Taxonomy of recommender systems was created and five commonly used e-commerce
1999		recommender application models described.
Ansari et al.,	Lab	Internet recommender systems can model customer preferences by using five types of
2000	simulation	information: a person's expressed preferences, preferences of other consumers, expert
		evaluations, item characteristics, and individual characteristics.
Haubl and	Lab	Recommender systems may improve individuals' choice outcomes in online shopping
Trifts, 2000	experiment	decision.
Herlocker,	Lab	Recommender systems have not been used in high-risk decision-making because of a
2000	simulation	lack of transparency.
Jiang, 2000	Lab	For naïve users, the confidence in the source of expertise, self-confidence, and the
	experiment	degree of discrepancy between the user's opinion and that of the expert system are all
		related to the persuasive power of the system.
Chiasson and	Case study	Recommender systems enable customers to cope with information complexity and
Dexter, 2001		information overload.

Table 1. (continued)

Desharnais et	Lab	Agents can provide tailored product recommendations by filtering information on
al., 2002	simulation	behalf of their users and reduce the information overload; vendors could use for price
		negotiation with the customers.
Rashid et al.,	Lab	Learning techniques of recommender systems affects the user effort and prediction
2002	experiment	accuracy.
Cosley et al.,	Lab	Users rate consistently across rating scales. Users can be manipulated by the expert
2003	experiment	system prediction. Users can detect expert system predication manipulation.
Gershoff, et al.,	Lab	Consumers pay special attention to extreme opinion agreement when assessing
2003	experiment	recommendation agent diagnosticity. Positive extreme agreement is more influencial
		than negative.
Wang and	Lab	Consumers' initial trust not only directly influences their intention to adopt the
Benbasat, 2004	experiment	recommendation agents but has indirect effect via their enhanced perceived usefulness
		of the agents.
Swaminathan,	Lab	Category risk moderates the impact of recommendation agents on decision quality and
2003	experiment	product complexity moderates the role of recommendation agents on amount of search.
Benbasat, 2004	Lab	Different types of explanation in recommender systems increase different trusting
	experiment	beliefs.
Porntpitakpa,	Review	A high-credibility source is superior over a low-credibility on persuasion.
2004		
Huang et al.,	Lab	Combining product content information and historical customer transaction
2004	simulation	information achieved more accurate prediction and relevant recommendation than
		using only collaborative information.
Yoon and Lee,	Lab	The product category, display format and other peripheral information provided by the
2004	experiment	p2p recommender system affect the behaviors of the potential customers.
Wang and	Lab	Consumer trust in recommender agents is an integral factor influencing their successful
Benbasat, 2005	experiment	adoption.

The studies showed that recommender systems provide better decision support with less user effort in information search. However, the outcomes of recommender systems typically depend on the context of the decision, the presentation of the recommendation, characteristics of the user, the object of the recommendation, and the user perceptions regarding the recommender systems. Learning techniques used by different recommender systems can affect user effort and prediction accuracy (Rashid, Albert, Cosley, Lam, McNee, Konstan, and Riedl, 2002). Combining product content information and historical customer transaction information achieved better prediction accuracy and recommendation relevance than using only collaborative information (Huang et al., 2004). Product complexity moderates the role of recommender systems on amount of search in online shopping by reducing the amount of search when the number of attributes used to describe a product is fewer (Swaminathan, 2003).

Given the benefits and increased use of recommender systems in the market place, it is necessary to study how different recommender systems influence customer preferences by comparing the persuasiveness of different recommender systems and probing the reasons behind it.

Evolution of Internet-Based Electronic Commerce

Since the introduction of the Internet in 1969 and the development of the World Wide Web in the early 1990's, the growing number of interconnected computers worldwide has provided Internet users with the potential to access an increasing abundance of information on a variety of subjects and providers. Internet usage has expanded from early academic research and technical development to both business and home users. It is not only perceived as a system able to send and receive information, but also as a tool for conducting business. According to the latest GVU survey by the Graphic, Visualization, and Usability Center at Georgia Institute of Technology (1999), the frequency of Internet usage (including hours on Internet for work and fun) for individuals has increased significantly. Additionally, more and more organizations (Bottoms, 1995; Clancy, 1995; Tetzeli, 1994) are using the Internet for advertising and interacting with customers.

Broader business use of the Internet became possible in the early 1990s through the efforts of a private consortium called the Commercial Internet Exchange Association (CIX) (Cross, 1994). The CIX persuaded the Internet community that commercialism was essential to its further development (Cross, 1994). Following this and the growing number of new Internet users, companies immediately began to establish a commercial presence on the Internet. The most prevalent method used was the development of a home page via a web site. According to a study conducted by Liu et. al. (1997), companies commonly include five elements on their web site: 1) information about products and services, 2) a company overview, 3) a customer feedback form, 4) a "what's new" area, and 5) an overview of financial information.

Encouraged by the promise of reaching new sales markets and enhancing operational efficiency, millions of international firms ranging in size from local, single proprietor businesses to multinational corporations have established their presence on the Internet via a web site. Many of these firms are already using their customized web site to support online business transactions involving both trading partners and direct customers. This breaks down the restriction of geographical consideration or lack of shelf space for the suppliers and brings about a broad range of customer choices. Yet new problems are encountered. It is hard for consumers to find their way in a large market place where so many suppliers and products are offered.

A filtering scheme like a recommender system is one solution that can propose relevant shops and products to a consumer based on customer preferences or the recommendation of like-minded people. The personalization of information provided by recommender systems can reduce consumer search effort and increase the purchase decision quality.

Development of Decision Support Systems and Intelligent Agents

Decision support systems (DSS) are computer technology solutions that can be used to support complex decision making and problem solving (Shim, Warkentin, Courtney, Power, Sharda, and Carlsson, 2002). Since the early 1970s, DSS technology and applications have evolved significantly. Many technological and organizational developments have impacted this evolution.

DSS once utilized more limited database, modeling, and user interface functionality, but technological innovations have enabled far more powerful DSS applications. Beginning in about 1990, Inmon and Kimball actively promoted data-driven DSS that were built using relational database technologies. In the early 1990s, a major technology shift occurred from mainframe-based DSS to client/server-based DSS. In 1994, Data Base Management Systems (DBMS) venders began implementing Online Analytical Processing (OLAP) capabilities into their databases (Powell, 2001). In 1995, web-based and web-enabled DSS began impacting practitioners and academics interested in decision support technologies (Bhargava, and Power, 2001; Power, 2000).

Early DSS supported individual decision-makers and included the model-oriented DSS, expert systems, multidimentional analysis, query and reporting tools, OLAP, business intelligence, and executive information systems. Later, DSS technologies were applied to workgroups or teams, the most exotic being applied to virtual teams (Shim et al., 2002). The next decade of DSS included more active decision support (Keen, and Morton, 1987) because organizations were becoming more agile and flexible while information sources for decision support were becoming more easily accessible and overwhelming in amount.

Intelligent systems are the new technology platforms that could be an alternative solution to this problem of active decision support. Rather than stand-alone artificial intelligence, intelligent logic is now usually inherent in the processing of all decision support systems. Intelligent systems are built to fulfill two key purposes: 1) the screening, sifting and filtering of a growing overflow of data, information and knowledge, 2) the support of an effective and productive use of standard software (Shim et al., 2002). Intelligent agents address the problem of data screening by filtering data sources for user-defined search profiles, identifying and accessing relevant data, copying

the data, and organizing and storing it in a data warehouse.

For electronic commerce, intelligent agents can help automate a variety of activities, mostly time-consuming ones. Benefits include lower customer search effort, improved decision quality, and lower the transaction costs. Intelligent agents may be categorized as personalized, social, continuously running and semi-autonomous (Guttman, Mouksas, and Maes, 1998b) and make e-commerce more user-friendly, semiintelligent and human-like. These qualities are conducive for optimizing the whole buying experience and revolutionizing commerce (Jeffrey, 1997).

The normative model of consumer buying process can be described as a learning, information-processing and decision-making activity divided in several sequent steps: need recognition, information search, alternative valuation, purchase decision and after purchase evaluation (Bettman, 1979; Boyd, Walker, Mullins, and Larreche, 2002; Brassington, and Pettitt, 2003; Dibb, Simkin, Pride, and Ferrell, 2001; Jobber, 2001; Kotler, 2003; O'Keefe *et al.*, 1998). Most academics and practitioners agree that demographic, socio-economic, cultural, psychological and other personal factors, largely beyond the control and influence of the marketer, have a major effect on consumer buying behavior (Boyd *et al.*, 2002; Czinkota, and Kotabe, 2001; Dibb *et al.*, 2001; Harrell, and Frazier, 1999; Jobber, 2001; Solomon, and Stuart, 2003). Despite their incapacity to exercise any substantial influence on the above factors, marketers can use marketing tools to influence consumer buying behavior and the final outcome of buyer-seller interaction (Brassington *et al.*, 2003; Kotler, 2003).

With the expansion of virtual markets, considerable research effort has been focused on moderating the online buying and decision-making process (Joines, Scherer, and Scheufele, 2003; Liao, and Cheung, 2001; Liu, and Arnett, 2000; McKnight, Choudhury, and Kacmar, 2002; Miles, Howes, and Davis, 2000; O'Cass, and Fenech, 2003). Compared to the traditional marketing place, online shopping includes a new step of building trust or confidence (Lee, 2002; Liang, and Lai, 2002; Liebermann, and Stashevsky, 2002; McKnight *et al.*, 2002; Suh, and Han, 2002). Chaung et al. (2003) identified two groups of factors associated with online consumer behavior, uncontrollable factors (consumer characteristics and environmental influences) and controllable factors (product/service characteristics, medium characteristics, and merchant/intermediary characteristics).

As the customer decision-making process is increasingly information-intensive with more information and multiple information sources, online recommender systems can be used as online marketer's persuasion tools to influence need recognition, information search and evaluation. The recommender systems can provide inputs for the consumers' black box where information is processed before the final consumer's decision is made so as to help consumers make better purchasing decisions.

Online Recommender Systems

Recommender systems provide a type of mass customization that is becoming increasingly popular on the Internet. Although the idea of recommender systems dates back to Negroponte (1970) and Kay (1984), practical implementation of "intelligent agents" is relatively recent and has been fueled by the success of online companies such as Firefly.com and Amazon.com. Early uses, which were not on the Internet, included short-lived in-store kiosks at Blockbuster Video that recommend films on the basis of member past rental history (West, Ariely, Bellman, Bradlow, Huber, Johnson, Kahn, Little, and Schkade, 1999). Magnet (Levy, 1993) claimed to be the first intelligent agent for the Macintosh. Essentially a file manager, it dispatched files into the trash if a user mistyped a destination folder (Foner, 1993).

Recommender systems are agents of the sort used by Blockbuster. Using behavioral or preference information, they filter alternatives and make suggestions to a user. Internet search engines are examples of content-based recommender systems, as these retrieve documents by means of keywords. In one commonly used system, the frequency of a target word is used to assess document relevance, and the relative frequencies of words are used to assess document similarity (Salton, and Buckley, 1988). Recommender systems screen attractive alternatives for individual-level predictions that can be useful even if there are a few alternatives. Recommender systems are most suitable for searching for goods which people do not always have the ability or the means to evaluate (e.g., cars and computers). Most important, recommender systems are able to work well with much less information because most people are averse to answering too many questions before they get recommendations (Ansari et al., 2000).

A number of researchers have categorized recommender systems according to different standards (Ansari *et al.*, 2000; Detlor *et al.*, 2002; Grenci *et al.*, 2002; Guttman,

Moukas, and Maes, 1998a; Maes et al., 1999; Montaner, Lopez, and de la Rosa, 2003; Mouksa, Zacharia, Guttman, and Maes, 2000). Among them, Ansari et al. divide recommender systems into collaborative filtering and content filtering based on the information sources of the recommendation. A second common classification of recommendation systems is based on whether they are designed to help a consumer determine what to buy or from whom to buy. These categories are product and merchant brokering, respectively (Guttman et al., 1998a; Maes et al., 1999). Another group of researchers introduced four types of recommender systems, based on the type of reasoning process: collaborative filtering recommender systems, constraint-satisfaction recommender systems, rule-based recommender systems and data-mining recommender systems (Ansari et al., 2000; Detlor et al., 2002; Grenci et al., 2002; Maes et al., 1999; Montaner et al., 2003; Mouksa et al., 2000). The classification of recommender systems defined by Ansari et al. (2000) will be used in this study to compare the persuasiveness of their recommendations. Although different reasoning processes could be used in the background by different online stores, the interfaces presented to the user are of two major types: those that need user input and those that do not. Therefore, this study will only compare two types of recommender systems, those that use content filtering vs. collaborative filtering for eliciting persuasiveness. The content filtering recommender systems require customer input. The collaborative filtering recommender systems depend on datamining of existing data.

Collaborative-filtering recommender systems utilize the opinions of like-minded

people to generate product recommendation. These kinds of recommender systems first compare customer product ratings with those of other customers. After identifying the "nearest neighbors " which are peer customers with similar tastes, the recommender system selects the products rated highly by chosen similar others (i.e. nearest neighbors). Since the customers may not yet have rated the item, this approach possibly leads to serendipitous findings (Ansari et al., 2000; Maes et al., 1999). Collaborative filtering requires a large amount of data to facilitate the correlation of consumer purchasing patterns, and recommendations cannot be changed or influenced by expert opinion. Thus, collaborative filtering is appropriate when decisions are not of significant importance, and is used primarily for products that are influenced by word-of-mouth. Until now, most of the uses for collaborative filtering have been with CDs, books, and music (Charlet, 1998b).

Content filtering is based on a system of selection logic rules that a retailer determines. Its biggest advantage over collaborative filtering is the ability to control the marketing process by creating rules. For instance, if a shopper has reviewed information on a new computer several times, a rule can be put in place to offer the shopper an extra 5% off if he purchases today. Also, it allows experts (i.e. customer managers) to tailor offers to certain shopper segments in order to offer relevant products and to maximize cross-selling opportunities. However, the system is only as good as programmer writes the rules, which implies continued maintenance costs (Charlet, 1998a).

Figures 1 and 2 are examples of a collaborative filtering recommender system

used at Amazon.com. The recommendation will be built based on products rated by the customer for similar taste. The 1 to 5 star-Likert scale was used in this example. Figure 3 is an example of content filtering recommender system where the product recommendation will be based on customer input of product attributes.

The Amazon.com rating page prompts customers to rate items recently purchased. These ratings are used as input to a recommendation engine to help the customer find other items that she/he is likely to buy. Customers are asked to invest effort in rating in exchange for which they get more useful recommendations. In Figure 1, no item has been presented because the user is a new customer while in Figure 2, items purchased by the customer before have been presented and rated for the system to provide recommendations for new purchase.



Figure 1. New Customer Recommender Interface at Amazon.com



Figure 2. Customer with Rated Items for Recommendation Building

In Figure 3, the advisor feature at Dell.com allows customers to indicate their preferences category to search when purchasing a product.

18

Back Forward Sto	Befresh Home	Search Favorites Medi	e History Mail Print	»
Address 🕘 http://search.dell.com/inc	lex.aspx?s=gen&c=us&l=en&cs=&	ec=k		🖌 🄁 📀
DELL USA 🔤				^
	🔒 Mya	Account 📔 🛪 Premier Login	🐺 My Cart 📮 My Order Status	
BACK TO: USA				
Soarch				
Search				
Search Dell USA		🕜 Search Ti	ps MOST POPULAR SEARCHES:	
Step 1: Choose An Area	to Search:	Need help choosin	g? ▶ Dell Axim	
 All Dell.com 			Dell Services & Training	
🚫 Home & Home Office	🚫 State & Local Government	○ K-12 Institutions	OTHER SEARCH OPTIONS:	
🔘 Small Business	🚫 Federal Government	🚫 Higher Education Institution	s 🖄 My Saved Searches	
🚫 Medium & Large Business	🚫 Healthcare	🚫 Member Purchase Program	♦ Submit Search Feedback	
Step 2:	a keyword or a part number			
	a keyword or a part namber.			
			_	
Step 3: Search Within:		-		~
			Internet	
B				

Figure 3. Customer Product Attributes/Constraints Input at Dell.com

Persuasion Theories

Persuasion, a concept originated from psychology and communication (Reardon, 1981), is the attempts to change the behavior of at least one person through symbolic interaction. It is conscious and occurs when both (a) a threat to goals of a subject is observed and (b) the source and degree of this threat are sufficiently important to warrant the expenditure of effort involved in persuasion (Reardon, 1991). It involves guiding people toward the adoption of some behavior, belief, or attitude preferred by the persuader through reasoning or emotional appeals. It is not a selection but presents a case

for the adoption of a persuader-preferred mode of action, belief, or attitude. Persuasion is a bilateral, incremental activity that involves some strategizing. Therefore, although the goal of persuasion is to change attitudes and/or behavior, persuasion needs to be properly presented because most people are naturally protective of their views and their behaviors and can close their minds to change if persuasion is not addressed in an acceptable fashion.

There are two broad types of theories of persuasion. The first ones downplay the role of reasoning in human behavior and might be described as push-and-pull theories that describe people as torn between inconsistent cognitions, pushed by stimuli or pulled by rewards. The second group of persuasion theory takes into account the ability of people to consider their actions and to reason through messages encouraging them to change during the course of persuasion (Reardon, 1991).

The second group of theories focuses on different aspects of persuasion. Bandura's social learning theory (Bandura, 1977) explains how patterns of behavior are acquired and how these patterns are influenced by both the self and external sources of influence. It demonstrates how people learn from direct and vicarious experience. The elaboration likelihood model (ELM) (Petty, and Cacioppo, 1986) explains why certain aspects of persuasive messages are sometimes more powerful than others in bringing about behavioral changes. The impact of persuasive elements depends on the information-processing route used by the persuadee. Communication/persuasion matrix model demonstrates how factors such as source, message, receiver, channel, and destination variables can be manipulated to evoke the 12 steps needed to bring about persuasion (McGuire, 1985). The ELM can be seen as a complement to McGuire's model, because it informs the persuader as to the priority he or she might want to attribute to factors affecting the central route versus the peripheral route.

The Elaboration Likelihood Model (ELM) (Petty *et al.*, 1986) clarified the difference between content and other related cues in determining the credibility of the messages. The ELM argues that persuasion in a communication can be achieved via one of two routes: central or peripheral. The central route is closely related to delivering strong, valid messages, so it is related to the content of information and its deliberate cognitive arguments around information content. In the absence of a central route, the peripheral route is used. It is based on social and affective cues rather than on the basis of message context. The examples of peripheral routes are the social relationship to the source and the social context.

Jaccard's study of expert systems advice acceptance provides another model to understand the process of persuasion communication (Jaccard, 1981). He proposed three variables as immediate psychological determinants for belief change: discrepancy of judgments, self-confidence and the confidence in the source of system knowledge. According to Jaccard, a low discrepancy, lower self-confidence, and a higher confidence in the source should lead to a higher acceptance of the advice given by the expert system.

The Appropriateness-Consistency-Effectiveness (ACE) model applies when the persuader is hoping to encourage the persuadee to reason about his or her behavior

(Reardon, 1981; Reardon, 1987). It includes certain stimulus inducements that can operate to encourage the adoption of a new behavior and suggests three categories of inducements. The ACE model can provide insight into how messages might be developed that encourage the central route processing of information described in ELM. This model can also provide some assistance in message formulation directed at evoking liking or interest, comprehension, skill acquisition, and yielding. While the ELM focuses on how persuasive messages are received, the ACE model focuses on the types of messages likely to be persuasive.

As mentioned before, persuasion is a complex activity. Factors influencing the persuasiveness of a message can be tracked to the message itself, the persuader and the persuadee. How the source of the message is perceived, gender and personality of the persuader, and the rapport with the persuadee are the persuader factors. Message variables include order of arguments, evidence, style of presentation, and salience of issues. The persuadee's gender, emotional dispositions, cognitive complexity, expectations, and schemata contribute to the persuasion outcome (Reardon, 1991).

Studies have also identified several factors specific to persuasiveness of recommender systems in electronic commerce, such as product complexity, consumer knowledge, category risk (Swaminathan, 2003), items characteristics, individual characteristics, expert evaluations, preferences of other consumers, and a person's expressed preferences (Ansari et al., 2000). In this study, we will focus on the product complexity factor and its impact on recommender system persuasiveness.

Research Questions

The research on online recommender systems focuses on two aspects: the impact of recommender systems on both the quality and efficiency of consumer decision making in an online shopping environment (i.e., how good a choice the consumer makes given the set of available products and how much effort he or she must expend to make a decision) and the impact on consumer behavior of a recommender system that elicits limited preference information before making a personalized product recommendation (Haubl *et al.*, 2000). No published comparison within or across two types of recommender systems has been done. This study conducted the comparison of two different recommender systems, the content-based filtering and collaborative filtering, to study their persuasiveness and the possible reasons for the differences.

The study tried to offer tentative answers specific to three questions:

- 1. Are there differences between the persuasiveness of the recommendation given by the two online recommender systems?
- 2. What are the factors causing the differences in the persuasiveness?
- 3. What is the role of product complexity on the persuasiveness difference?

This study examined which type of recommender system is more persuasive in the context of e-commerce and why. The answer will be important to designers and owners of online stores in considering what recommender systems to use for certain products. The higher level of recommender system persuasiveness will also lead to higher acceptance by customers (Jiang et al., 2000).

Research Methodology

The research methodology section consists of a brief discussion of the research sample, the research instrument, and data analysis. A more in-depth discussion of the methodology is provided in Chapter III.

Research Sample and Design

The research sample of this study consists of a convenience sample of about one hundred undergraduate students from a major southern university. They were recruited to participate in a web-based experiment conducted in a computer lab. Respondents were told to search for two different products (high vs. low complexity) with two websites (www.amazon.com and www.activebuyersguide.com) using different recommender systems. To control for participant differences, a within-subject design with repeated measures was used. Each participant was asked to use two recommender systems, one at a time, to purchase two products. As an incentive for participating, the respondents were told that they will be given extra credits for taking part in the experiment. On completing each product search, participants filled out a survey with questions intended to measure appropriateness, consistency, effectiveness, and their perception of the recommender systems persuasiveness. Other data such as age, gender, consumer product involvement with each product and experiences with online shopping were collected for analysis. A pilot study was conducted for testing of the instrument before data collection from the main sample.
Covariate

Product involvement is a factor that may influence consumers' cognitive and behavioral response-including memory, attention, processing, search, brand commitment, satisfaction, early adoption, and opinion leadership (Laaksonen, 1994). According to Zaichowsky (1985), product involvement is the relevance that individuals perceived in a product, according to their inherent values, interests, and needs. If consumers feel that a product is important to their life, they will be more likely to process the communication along the central route and expend more effort processing or understanding the message. Product involvement represents one of the main motivations to process communications.

The concept of product involvement has been receiving more research attention recently to study consumers' response to Internet advertising (Ahn, and Edwards, 2002; Cho, and Leckenby, 1999; Karson, and Korgaonkar, 2001; Laczniak, Kempf, and Muehling, 1999; Yoo, and Stout, 2001). Studies also indicated that consumer's involvement influences consumer flow experience and subsequent exploratory search behavior (Hoffman, and Novark, 1996; Koufaris, Kambil, and LaBarbera, 2001-2002; Quester, and Smart, 1998). Involved consumers are more diligent in examining and detecting differences among brands (Zaichowsky, 1985). Highly involved consumers use a central route in information process. Cho (1999) found support that product involvement influenced subjective motivation to processing web advertising content. Product involvement also affect the relationship between online advertisement interactivity and consumer comprehension (Macias, 2003). It was also pointed out that product involvement is not constant for a product or type of product but varies for each customer (Koufaris et al., 2001-2002).

In the context of recommender systems, it can be assumed that consumers of different product involvement may require different information search need and explanation details given by the recommender systems. In this study, product involvement was set as a covariate to investigate its impact on persuasiveness of the recommendation as well as its possible interactions with types of recommender systems and product complexity.

Research Instrument

In this study, four dependent variables have to be measured. They are effectiveness, appropriateness, consistency, and persuasiveness. The persuasiveness and appropriateness scales used an existing scale in marketing, the persuasive disclosure inventory (PDI) for judgment of ads. According to McGuire (1969), the theoretical components of the PDI measures are ethos, pathos, and logos. Ethos refers to persuasive appeals that concentrate on the source rather than the message. Studies of advertising effects that have examined emotional or affective appeals fall within the definition of pathos. A logos appeal provides evidence or information about a concept from which a consumer can form beliefs (1994). Pathos and logos have been viewed by some as different ends of a continuum that considers the message. Feltham (1994) suggests that the subscales be used as individual message facets. The reliability estimates for the Pathos, Ethos, and Logos subscales were .83, .79, and .89 respectively. There is another

persuasiveness scale developed by Keller and Block (Keller, and Block, 1997) to measure the effectiveness of a brochure to change the attitude of a person toward some topic. The topics focused on in the scales were health-related and the emphasis was on gauging the reader's expressed intention to comply with the behavior suggested in the brochure. The scale had a high reliability (0.84) but no examination of the validity was reported by Keller and Block (1997). Compared to McGuires PDI, the brochure persuasiveness scale lacks face validity. The logos items were used to measure appropriateness and the ethos and pathos items were used to measure persuasiveness.

The scale for effectiveness used was the sales presentation effectiveness measurement developed by Behrman and Perreault (1984). The original scale was a sixitem, five-point Likert-type scale purporting to measure salespeople's self-evaluation of the effectiveness of a sales presentation. Reliability testing by Strutton and Lumpkin (1994) revealed a Cronbach's alpha of 0.8. Both discriminant and convergent validity were reported by Behrman and Perreault (1984). Strutton and Lumpkin (1994) also proved the unidimensionality of the underlying data using maximum likelihood confirmatory factor analysis.

The questions used for consistency was from a study by Morris (1994) where items were used to test whether the recommendations from different databases are consistent with a user's existing knowledge. The wording of the questions was adapted to the Purdue Consistency Testing Questionnaire on web site concept consistency (Ozok, 1997). Product complexity measurement used the five-item, five-point semantic differential summated ratings scale measuring the perceived complexity of a product developed by McCabe (1987). The scale has a reported reliability of 0.80.

A variety of involvement frameworks have been proposed to guide advertising research. Most of the frameworks focus on the effects of some form of situational involvement on advertising effectiveness measures, such as processing of ad content or attitude toward the ad. Those scales are classified according to type of involvement and object of involvement (Day, Stafford, and Camacho, 1995). In this study, the four-item, eight-point Likert-type summated product involvement scales by Zinkhan and Locander (1988) was used. The original scale was developed to measure the degree of involvement a consumer has with calculators. The scale showed a reliability of 0.9 in the study.

Data Analysis

Data analysis consists of three phases: 1) pretest of the research instrument, 2) data collection and assessment of its measurement properties, and 3) hypotheses testing. To ensure the instrument has both face and content validity, the initial questionnaire was pre-tested by a smaller sample from a different group of students. Appropriate changes were made in accordance with pre-test results. The instrument was then administered to the identified sample.

First, the validity and reliability of all scales used was tested using factor analysis. The next phase is to test the research hypotheses. MANCOVA procedures will be employed to guard against type I error. In addition, because four dependent variables (appropriateness, consistency, effectiveness, and appropriateness) in this study are conceptually related according to Reardon's persuasion theory, the MANCOVA procedure is more suitable for this type of analysis that controlled correlations among dependent variables (Bary, and Maxwell, 1985). Product involvement will be used as a covariate to study its impact on persuasiveness and possible interactions with other factors. For main factors, differences of means related to the hypotheses were tested for statistical significance if the main effects show statistical significance. Univariate ANOVA analysis was used to test the sub-hypotheses 1-3 and ANCOVA was used to test hypotheses 4a, 4b, and 4c.

Implications of the Research

Most previous research on recommender systems has focused on the statistical accuracy of the algorithms driving the systems, with little emphasis on interface issues and the user's perspective. Recently, studies have begun to focus on the end-user side of the issue. Researchers have examined the impact of transparency (Sinha *et al.*, 2002), explanation (Herlocker, Konstan, and Riedl, 2000; Schafer, Konstan, and Riedl, 1999a; Ye *et al.*, 1995), trust, category risk, consumer knowledge, and product complexity (Swaminathan, 2003) on acceptance of recommendations given by recommender systems and its further influence on electronic commerce and online shopping. The investigation of persuasiveness provides another perspective to understand the online recommender systems to improve online recommendation targeting and to lead product searching into actual

purchasing.

Studies have been conducted to compare recommendations from recommender systems with human recommendations (Sinha, and Swearingen, 2001) and other expert systems (Morris, 1994). No research has been done to compare two recommender systems. Although there have been many algorithms used in recommender systems, collaborative-filtering and content-filtering are the two major types of algorithms that can be easily noticed by online consumers. The outcome of the research can help online vendors gauge and combine different algorithms for products of different complexity for better recommendation acceptance. Recommender systems can play the role of virtual advice-giving salespersons in an online context.

The use of persuasiveness theory brought in new insight to the end-user computing research in information systems. The Technology Acceptance Model (TAM) has been widely used to study different kinds of systems with additional constructs added. The persuasiveness theory is an alternative and concise model that can help explain the factors affecting the behavior of potential customers.

Product involvement is a well-researched factor in advertising study and has been investigated for its impact on web-based consumers. The introduction of this construct in the information system related research will provide a new application for this marketing concept and can bring novice understandings of the e-commerce related systems from a consumer's perspective.

Finally, the design of the research offered a within-subject design technique

which is common in other disciplines but not so frequently used in information systems research. This method is highly sensitive in detecting statistically significant differences and allow for the use of much smaller sample sizes than necessary for between-subject designs (Cozby, 1989). Similarly, research in information systems tends to focus on development of new measurement scales for each study, to increase the validity of the instrument. The reuse of established scales tends to be neglected (Straub, 1989). This is contrary to the scientific principle of validation through repeated studies with the same instrument, and necessitates the use of large sample sizes as a diverse sample frame for each study. As a result, some worthwhile studies may not be performed due to the inability to secure large enough samples for the study.

The limitations of this study include the use of student samples, validity of scales, the carryover effect in the experiment design, and the use of existing recommender systems. Details will be given in the later discussion and summary chapter.

Items used in the questionnaire are listed in Appendix A.

Organization of the Study

This dissertation is organized into five chapters with appendixes. Chapter I provides an overview of the concepts of electronic commerce, decision support systems, intelligent agent, recommender systems, and persuasiveness theory. The research problems, significance of the research, research methodology, research contribution, and research implications are also included in this chapter. Chapter II provides an overview of the research relevant to persuasiveness, electronic commerce issues, factors impacting recommendation persuasiveness, and the research hypotheses for this study. Chapter III explains the methodology for sample selection, statistical procedures, and data analysis. Chapter IV presents the data analysis and results. Chapter V presents a summary of research findings, implications of the study, and directions for future research.

CHAPTER II

RECOMMENDER SYSTEMS AND HYPOTHESES DEVELOPMENT

This chapter presents a review of the research that comprises the theoretical basis for the current study. The chapter is divided into three parts: 1) an overview of the relevant research about the evolution of recommender systems in e-commerce, 2) a summary of the research about persuasiveness, and 3) the product complexity factor that is relevant to this study.

Recommender Systems in E-commerce

One of the earliest definitions of electronic commerce states that electronic commerce is a new way of conducting business characterized by companies and their customers performing electronic transactions through computer networks (Cronin, 1994). Based on this definition, technologies including Electronic Data Interchange (EDI), Electronic File Transfer (EFT), E-mail, facsimile (FAX), bar coding symbol technology, and enterprise messaging and file transfer have been linked to the development of electronic commerce within the business sector (Pyle, 1996). Yet it was not until the growing commercial use of the Internet, particularly the web subset of ITCP/IP, an interest in the benefit of conducting business electronically developed (Zwass, 1996). In this chapter, the focus is on the narrower definition of e-commerce, especially the commercial activities conducted over the Internet (Hake, 1999). Electronic commerce offers opportunities for businesses to make faster, cheaper, more personalized and more flexible ways to interact with both customers and suppliers. According to the nature of its transactions, e-commerce can be categorised into the following types: business-tobusiness(B2B), business-to-consumer(B2C), consumer-to-consumer(C2C), consumer-tobusiness(C2B), nonbusiness ecommerce (use of the Internet by nonbusiness organizations such as academic institutions or government agencies to reduce expenses or improve services), and intrabusiness e-commerce (Turban, Lee, King, and Chung, 2004). Currently, most recommender systems are either B2C or B2B. B2C will be the focus of this study.

B2C mainly refers to online retailing transactions with individual customers, where shoppers can conduct transactions through the webpage of a company. B2C ecommerce is becoming more widespread as more people are recognizing its convenience and capability to offer quick responses to requests as more products/services become available (Murch, and Johnson, 1999). He et al. (2003) stated that recommender systems can act as mediators in five of the stages of the consumer buying behavior (CBB) model. These stages are:1) need identification, 2) product brokering, 3) buyer coalition formation, 4) merchant brokering and 5) negotiation. This study will discuss the product brokering mediated by recommender systems where the systems assist the user about what product to buy to satisfy his or her need for some product or service.

He et al. (2003) divided e-commerce systems into two generations. In the first generation, buyers are generally humans who typically browse through a catalog of well-

defined commodities (e.g., flights, books, computer components) and make fixed price purchases (often by credit card). The second generation e-commerce systems have a greater degree of automation on both the buyer's and the seller's side. Commerce becomes much more dynamic, personalized, and context sensitive in the context of webbased transactions. These changes can be of benefit to both the buyers and the sellers. From the buyers' perspective, it is desirable to have software that could search all the available outlets to find the most suitable one for purchasing the chosen good (e.g., the one that offers the cheapest price, the highest quality, or the fastest delivery time) and that could then go through the process of actually purchasing the good, paying for it, and arranging delivery at an appropriate time. From the sellers' perspective, it is desirable to have software that could vary its offering depending on the customer it is dealing with, what its competitors are doing, and the current state of its business. For example, the software agent can offer discounts or special offers to particular target groups. It can continuously monitor competitiors's prices and make sure its own price is competitive. The software agent can also reduce the product price to try and increase demand if the store has plenty of a particular item in stock.

Software agents like recommender systems distinguish the second generation of e-commerce applications. According to Shafer et al. (1999b), recommender systems can enhance e-commerce sales in three ways: 1) converting browsers into buyers, 2) increasing cross-sell, and 3) building loyalty. Recommender systems can help consumers find products they wish to purchase instead of pure searching. Additional products suggested during the search process could increase the average order size. Recommender systems improve loyalty by creating a value-added relationship between the site and the customer. When sites invest in learning customers using recommender systems to match customer need, consumers repay these sites by returning to the ones that best match their need. The more a customer uses the recommender system by teaching it what he wants, the more loyal he or she is to the site. Gaining customer loyalty is an essential business strategy where a site's competitors are only a click or two away (Reichheld, 1993; Reichheld, and Sasser, 1990).

Characteristics of Recommender Systems

Because recommender systems are software agents that can use behavioral or preference information to filter alternatives and make suggestions to a user (Ansari et al., 2000), it should have the basic characteristics of an intelligent agent. That is the capability of flexible autonomous actions to meet its design objectives. To achieve this, the software must exhibit these properties: reactivity, pro-activeness, autonomy, and social ability (Ansari et al., 2000; He et al., 2003).

Reactivity is the capability to respond appropriately to the prevailing circumstances in dynamic and unpredicted environments (Wooldridge, and Jennings, 1995). In other words, the recommender systems should be able to perceive and respond in a timely fashion to changes that occur in their environment in order to satisfy their design objectives. Pro-activity refers to the ability to act in anticipation of future goals so that the objectives of the owner are met. This way, when the environment changes, recommender systems can recognize opportunities and take the initiative if they are to produce meaningful results. The challenge to the agent designer is to effectively integrate goal-directed and reactive behavior (Rudowsky, 2004). Autonomy means the recommender systems must be able to make decisions about what actions to take without constantly referring back to their users (He et al., 2003). Sociability refers to the ability to interact with other agents or humans through negotiation and cooperation to satisfy design objectives (Rudowsky, 2004)

Approaches Used in Recommender Systems

Many different approaches have been used to solve the basic problems of making accurate and efficient recommender systems. Many of the technologies used in the recommender systems studied are fairly simple database queries. Automatic recommender systems, however, use a wide range of techniques, ranging from nearest neighbor algorithms to Bayesian analysis. The worst-case performance of many of these algorithms is known to be poor. However, many of the algorithms have been tuned to use heuristics that are particularly efficient on the types of data that occur in practice (Schafer et al., 1999b).

The earliest recommenders used nearest-neighbor collaborative filtering algorithms (Resnick, Iacovou, Suchak, Bergstrom, and Riedl, 1994; Schafer *et al.*, 1999b; Shardanand, and Maes, 1995). Nearest neighbor algorithms are based on computing the distance between consumers based on their preference history. Predictions of how much a consumer will like a product are computed by taking the weighted average of the opinions of a set of nearest neighbors for that product. Neighbors who have expressed no opinion on the product in question are ignored. Opinions should be scaled to adjust for differences in rating tendencies between users (Herlocker, Konstan, Borchers, and Riedl, 1999). Nearest neighbor algorithms have the advantage of being able to rapidly incorporate the most up-to-date information, but the search for neighbors is slow in large databases. Practical algorithms use heuristics to search for good neighbors and may use opportunistic sampling when faced with very large populations.

Bayesian networks create a model based on a training set with a decision tree at each node and edges representing consumer information. The model can be built off-line over a matter of hours or days. The resulting model is very small, very fast, and essentially as accurate as nearest neighbor methods (Breese, Heckerman, and Kadie, 1998). Bayesian networks may prove practical for environments in which knowledge of consumer preferences changes slowly with respect to the time needed to build the model, but are not suitable for environments in which consumer preference models must be updated rapidly or frequently (Schafer et al., 1999b).

Clustering techniques work by identifying groups of consumers who appear to have similar preferences. Once the clusters are created, predictions for an individual can be made by averaging the opinions of the other consumers in that cluster. Some clustering techniques represent each consumer with partial participation in several clusters. The prediction is then an average across the clusters, weighted by degree of participation. Clustering techniques usually produce less-personal recommendations than other methods, and in some cases, the clusters have worse accuracy than nearest neighbor algorithms (Breese et al., 1998). Once the clustering is complete, however, performance can be very good, since the size of the group that must be analyzed is much smaller. Clustering techniques can also be applied as a "first step" for shrinking the candidate set in a nearest neighbor algorithm or for distributing nearest-neighbor computation across several recommender engines. While dividing the population into clusters may hurt the accuracy or recommendations to users near the fringes of their assigned cluster, preclustering may be a worthwhile trade-off between accuracy and throughput (Schafer et al., 1999b).

Information filtering and information retrieval involve selecting text items that a user may be interested in reading based on the presence or absence of keywords in the text items. The users can explicitly enter the keywords, or the keywords can be inferred from the items that users have found interesting in the past. Information filtering or information retrieval systems are often used in search systems on e-commerce sites to help consumers find specific products in which they are interested (Schafer et al., 1999b). These systems have some features in common with recommender systems in that both systems produce lists of suggestions for a user. However, the more the system provides direct responses to syntactic user queries the less it feels like a recommender system to the user. Information filtering systems that notify users when interesting items are for sale are more like recommender systems, especially if part of the selection process involves attributes that are not under direct control of the user, such as whether other users have

liked the item.

Classifiers are general computational models for assigning a category to an input. The inputs may be vectors of features for the items being classified or data about relationships among the items. The categories are a domain-specific classification such as malignant/benign for tumor classification, approve/reject for credit requests, or intruder/authorized for security checks. One way to build a recommender system using a classifier is to use information about a product and a customer as the input and to have the output category represent how strongly to recommend the product to the customer. Classifiers may be implemented using many different machine-learning strategies including rule induction, neural networks, and Bayesian networks. In each case, the classifier is trained using a training set in which ground truth classifications are available. It can then be applied to classify new items for which the ground truth is not available. If subsequent ground truths become available, the classifier may be retrained over time (Schafer et al., 1999b).

Classifiers have been quite successful in a variety of domains ranging from the identification of fraud and credit risks in financial transactions to medical diagnosis to intrusion detection. Basu et al.(1998) built a hybrid implementation induction-learned feature-vector classification of movies and compared the classification with nearest-neighbor recommendation. They found that the classifiers did not perform as well as nearest neighbor, but combining the two added values over nearest-neighbor alone.

Association rules have been used for many years in merchandising, both to

analyze patterns of preference across products and to recommend products to consumers based on other products they have selected. An association rule expresses the relationship that one product is often purchased along with other products. The number of possible association rules grows exponentially with the number of products in a rule, but constraints on confidence and support, combined with algorithms that build association rules with itemsets of n items from rules with n-1 itemsets, reduce the effective search space. Association rules can form a very compact representation of preference data that may improve efficiency of storage as well as performance. They are more commonly used for larger populations rather than for individual consumers, and they, like other learning methods that first build and then apply models, are less suitable for applications where knowledge of preferences changes rapidly. Association rules have been particularly successful in broad applications such as shelf layout in retail stores. By contrast, recommender systems based on nearest neighbor techniques are easier to implement for personal recommendation in a domain where consumer opinions are frequently added, such as online retail (Schafer et al., 1999b).

Horting is a graph-based technique in which nodes are consumers, and edges between nodes indicate the degree of similarity between two consumers (Wolf, Aggarwal, Wu, and Yu, 1999). Predictions are produced by walking the graph to nearby nodes and combining the opinions of the nearby users. Horting differs from collaborative filtering as the graph may be walked through other consumers who have not rated the product in question, thus exploring transitive relationships that traditional CF algorithms do not consider.

Most of the online stores consider the algorithms they use to be proprietary. Individual recommender systems may actually use a combination of these algorithms while still presenting the same interface to the user. For this reason, this study concentrates on the two types of recommender systems that require different consumer input at the interface instead of the specific technologies used.

Types of Product Brokering Recommender Systems

To be able to make recommendations, the recommender system needs information from five information sources: 1) expressed preferences or choices of a person among alternative products, 2) preferences for product attributes, 3) preferences or choices of other people, 4) expert judgments, and 5) individual characteristics that may predict preferences (Ansari et al., 2000). A good recommender system should be able to use any or all of these five types of information to make better recommendations.

Based on the type of information needed and the presentation to consumers, recommender systems have been classified into three types. Those are the needbased/rule-based filtering, collaborative filtering and constraint-based/attributesbased/feature-based filtering (Ansari et al., 2000; He et al., 2003; Maes et al., 1999).

Need-based filtering identifies the needs of a customer (i.e. the intended use of a product, item location, the price range or the date limit and so on) and then recommends a product that meets those needs. Such a recommender system is expertise-driven, because a set of rules serve to interpret the specific information or intentions of the customer into

a recommendation product configuration (Grenci *et al.*, 2002). The focus in need-based filtering is on the customer's intended use of a product, assuming the customer already knows the intended use of the product. This need-based filtering is also termed as rule-based filtering (Maes et al., 1999). For example, eBay guides a user to select the products by narrowing down the range of the possibilities based on the constraints the user gives and provides a list of the desired products that satisfy the constraints given by the user.

Collaborative filtering identifies how customers are grouped together with likeminded people, and explains that the product recommendations are based on the opinions of such people, assuming that they would like to buy similar products. Collaborative filtering predicts preferences of a person as a weighted sum of other people's preferences, in which the weights are proportional to correlations over a common set of items evaluated by two people (Ansari et al., 2000). For example, in CDNOW (www.cdnow.com), users are notified about the CDs or movies that are popular with other users with similar preferences.

Collaborative filtering algorithms have several limitations. First, when data are sparse, the correlations (weights) are based on few common items and therefore are unreliable. Second, collaborative filtering algorithms can be used only when preference data for an item already exists in the database. In other words, the system cannot handle queries about new items. Third, these methods use ad hoc prediction algorithms, which are not based on statistical models. Therefore, the uncertainty of the recommendation is high. This may be less important for low-risk purchases such as movies and compact discs but can be very important when the stakes are higher for a consumer or company. Fourth, collaborative filtering systems do not explicitly incorporate attribute information, though they are bootstrapped by creating "virtual users" who represent particular tastes. Finally, collaborative filtering methods are correlational. This feature provides little explanation for a recommendation and can be important for building trust and enhancing customer loyalty (Desharnais *et al.*, 2002; Wang *et al.*, 2003).

Attribute-based filtering recommender systems allow recommendations by first asking customers about their preferred levels of certain product attributes because these attributes are critical in product decision making as judged by the recommender systems based on their product expertise. Then, the system recommends a list of products that satisfy all customer hard constraints and is ordered by how well the products satisfy soft constraints. This allows recommendations for entirely new items but does not necessarily incorporate the information in preference similarity across individuals (Sarwar, Konstan, Borchers, Herlocker, Miller, and Riedl, 1998; Shardanand *et al.*, 1995). Similar to collaborative filtering, attribute-based filtering cannot make recommendations for people who provide no preference information. Systems that use neural networks often have difficulty providing explanations for recommendations (Ansari et al., 2000). Table 2 shows a comparison of recommender systems using different filtering algorithms.

Different types of recommender systems can have different implicit or explicit explanations regarding their reasoning process (i.e., their different ways to generate product recommendations). These can lead to different levels of recommendation persuasiveness. Wang and Benbasat (2003) show that different types of explanations in recommender systems increase different trusting beliefs of the consumer to recommendation generated. Rashid (2002) found that learning techniques of recommender systems affect the user effort and prediction accuracy. Furthermore, recommender systems have not been used in high-risk decision-making because of a lack of transparency in recommendation generation.

Table 2. Comparison of Filtering Algorithms Used by Recommender Systems(He et al., 2003)

	Feature-based	Collaborative	Constraint-based
Users' needs	Known	Unknown	Some known
Input	Product feature	User profiles	Functionality
Interaction	Medium	Few	Medium
Output	Goods with required features	Suggestions of goods to buy	Goods with required functionality
Suitability	Most goods	Books, CDs, etc.	Most goods

Persuasion and the Acceptance-Consistency-Effectiveness (ACE) Model

In its most general definition, persuasion is the process of attempting to change the behavior, belief, or attitude of the persuadee(s) toward a persuader-preferred mode of action, belief, or attitude (Reardon, 1991). In the context of using recommender systems to automate or assist customer product decision making, the term *persuasive* refers to the extent to which customers are moved or influenced by the recommender systems' reasoning to a belief that the recommended product best fits their personal needs (Komiak *et al.*, 2004).

In this study, the ACE model is used to compare the persuasiveness of two different recommender systems for several reasons. First, the ACE persuasion model provides a perspective different from adoption and acceptance of recommendations generated by recommender systems in e-commerce. In management information system research, the topic of user beliefs and attitudes change has been studied extensively (Barki, and Hartwick, 1994; Baroudi, Olson, and Ives, 1986; Fishbein, and Ajzen, 1975; Ginzberg, 1981; Hartwick, and Barki, 1994; Pare, and Elam, 1995). Especially user satisfaction and perceived usefulness are believed to be important factors for system acceptance, adoption, and system use (Davis, 1989; Davis, 1993; Davis, Bagozzi, and Warshaaw, 1989; Galletta, and Lederer, 1989; Gatain, 1994; Igbaria, and Machman, 1990; Igbaria, Schiffman, and Wieckowski, 1994; Iivari, and Ervasti, 1994a; Ives, Olson, and Baroudi, 1983; Keil, Beranek, and Konsynski, 1995).

For recommender systems, users cannot prove the correctness of the recommendation (whether the delivered products satisfy consumer requirement and needs) before they receive the products. With the convenience of the Internet, customers can search multiple sites for recommendations at almost no cost. Therefore, user acceptance of a recommendation is not supported by the actual satisfaction but by user beliefs about the recommendation argumentation given by the recommender system. This indicates the persuasiveness of the system. Kottemann et al. (1994) show that redundant

what-if help increases users' confidence in the decision, and Davis et al.(1994) point out that irrelevant information in an information system weakens performance but increases the user confidence in decision making. Similar results were reported by Aldag and Power (1986) and Will (1992).

Persuasiveness reflects the power of the recommender systems to convince the customer. Persuasion process is a complex activity involving the sources, the media, and the receiver. The customer confidence in the recommender system, the customer self-confidence in decisions, and the degree of discrepancy between the customer opinions and the advising system opinions are all related to the persuasive power of the system (Jiang et al., 2000). The three variables have been described as "the immediate psychological determinants of belief change" (Jaccard, 1981). From the stand point of a recommender system, the source is the recommender system, the message is the recommendation, and the receiver is the consumer. The objective of a recommender system is to display recommendations to consumers and thus support and influence their final decision. Persuasion theory can be used to examine the persuasiveness of recommendations from different recommender systems.

The second reason for using the ACE model is its focus on recommendation messages rather than the consumer decision making process (Reardon, 1991). There are many factors affecting the persuasiveness of a recommendation from a recommender system. The study of persuasion processes should investigate all the elements of the communication process: source (e.g., expertise, trustworthiness, sex), message (e.g., tone, discrepancy, order effects, fear appeals, argument quality), and audience (e.g., selfesteem, intelligence, cognitive complexity, self-efficacy, perceived difficulty) (Areni, Ferrell, and Wilcox, 2000; Fishbein *et al.*, 1975; Fishbein, Hennessy, Yzer, and Douglas, 2003; Yzer, Hennessy, and Fishbein, 2004). It would be impossible to identify and measure all of the potential relevant features of the recommender system, kinds/formats of recommendation, and types of users in a given study.

The ELM model focuses on how persuasive messages are received. It predicts the likelihood that the persuadee will attend to the persuasive appeal, attempt to access relevant associations and experiences from memory, scrutinize and evaluate the information in light of the associations drawn from memory, draw references about the merits of the arguments, and derive an attitude toward the recommendation (Petty, Kasmer, Haugtvedt, and Cacioppo, 1987). When elaboration is needed, a person is more likely to use the central route to persuasion. When elaboration is not critical, the peripheral route is preferred. There are a number of variables that might affect the likelihood of elaboration, and a variable can play different roles in different persuasion situations. When people are either unmotivated to evaluate or incapable of evaluating the true merits of the arguments presented, they base their judgments on simple cues.

A fairly large body of research in the consumer information-processing domain suggests that the persuasive impact of advertisements depends on the extent to which executional cues within the advertisement are compatible with consumers' likely elaboration of brand information in the advertisement. When advertisements target consumers who are high in motivation, ability, and opportunity, they are most effective when they contain rational executional cues that credibly demonstrate the benefits of the product compared with competitive offerings (Petty, Shumann, and Cacioppo, 1983). Such cues enable consumers to engage in issue-relevant thinking and evaluate the true merits of the brand. The potential pool of rational cues includes messages 1) ranging from factual to feeling based (Olney, Holbrook, and Batra, 1991; Stewart, 1999; Torson, and Page, 1988), 2) indicating how the product is different from competitive product offerings (Stewart, 1999), 3) demonstrating product superiority (Petty, and Cacioppo, 1984; Sewall, and Sarel, 1986), and 4) focusing on product (MacInnis, and Jaworski, 1989) attributes and benefits as opposed to the user (Maheswaran, and Sternthal, 1990; Malaviya, Kisielius, and Sternthal, 1996; Stewart, 1999). When consumers motivation, ability, and/or opportunity to process ad information are low, such consumer devotes limited effort to processing message content or lack of sufficient knowledge to interpret and understand attribute-based information. Prior research has shown that under these low-elaboration likelihood conditions, cues such as expert endorsers, similar endorsers, and attractive pictures enhanced persuasion. These cues include affectively based cues and Heuristic cues.

Affectively based cues include likeable sources (Chaiken, 1980; Kahle, and Homer, 1985; Petty, and Cacioppo, 1981; Petty *et al.*, 1983), drama (Deighton, Romer, and McQueen, 1989), warm appeal (Stayman, 1990), visually appealing pictures (Grossman, and Till, 1998; Mitchell, 1986; Mitchell, and Olson, 1981; Stewart, 1999), and likable music (Bierley, McSweeney, and Vannieuwlerk, 1985; Gorn, 1982; MacInnis, and Park, 1991). Prior research has found that such cues can induce positive feelings in viewers and positively influence ad and brand attitudes.

Heuristic cues are shortcuts that enable inferences about brand benefits or quality. For example, although consumers may be unable to discern whether a brand is of high quality, they may come to believe it is when it is advertised by a credible source (Craig, and McCann, 1978; Sternthal, Dholakia, and Leavitt, 1978), or someone knowledgeable about the product category (Chaiken, and Maheswaran, 1994; Yalch, and Yalch, 1984). Although consumers may be unable to diagnose technical features of the brand made in ad claims, they may make inferences about its benefits from such easily processed persuasion cues as ad-relevant pictures (Kahle *et al.*, 1985; Miniard, Bhatla, Lord, Dickson, and Unnava, 1991; Mitchell *et al.*, 1981) or relevant music.

Although the ELM has gained considerable recognition, its application has been limited. Because the theory groups diverse processes (e.g., more coginitively based heuristic persuasion evoked from heuristic cues versus affectively based processing evoked from affective cues) under the general "peripheral" routes to persuasion. It is possible that various conditions dictate when certain types of peripheral cues are more effective than others.

Anderson's (1971) information integration theory has been applied largely to persuasion situations by including factors involved in the persuasion process. According to Anderson, the response of an individual on a scale after a persuasive message is a weighted additive function of the premessage belief of the individual and the position of the source of the communication. It can be expressed mathematically as $R = W_0S_0 + W_IS_I$ where R = the belief of the person after being exposed to the message, S_0 = the "scale value" of initial opinion of the person, S_I = the scale value of the source position, and W_0 and W_I = weighting parameters reflecting the psychological importance of the initial belief of the individual and the source position. The biggest challenge of this model is the difficulties in determining estimates of W_i and S_i .

It is understood that the study of persuasion process is an arduous task. People do not always reason about the information they receive (Petty et al., 1987). Petty et al. (1987) found that it is neither adaptive nor possible for people to exert considerable mental effort in processing all of the persuasive communication to which they are exposed. When motivation and ability to scrutinize information are low, it is still possible for people to form a new attitude or change an old one, but they do so through simple association, inferences, or heuristics rather than exerting the mental effort involved in reasoning about alternatives.

Research found that users sometimes accept expert system advice without going through thorough examination of correctness or the application of a cognitively convincing argument. They are likely to make their judgment on peripheral cues if they are less motivated or unable to judge a message on its contents (Petty *et al.*, 1986). Others simply want to reduce the cognitive effort by accepting advice of an expert system, even when they do not like it (Todd, and Benbasat, 1992).

This can be supported by more general research on decision making. According to Shugan (1980), the amount of thought that a human decision maker devotes to making a particular choice depends largely on the degree of decision making difficulty. Thinking effort is positively related to both the complexity of the decision and the desired level of confidence in having made the best possible choice. It is inversely related to the difference in the decision maker preference between the available options. As a result, complex and important decisions are more costly in terms of cognitive effort than simple and routine decisions. Individuals often settle for less accurate decisions in return for reduction in effort (Buttman, Johnson, and Payne, 1990). This is particularly true when alternatives are numerous or difficult to compare (Payne, Bettman, and Johnson, 1993). The use of decision support may result in effort savings but not improved decision performance (Todd et al., 1992). Because feedback on effort expenditure tends to be immediate while feedback on accuracy is subject to delay and ambiguity, decision makers may be inclined to focus more on reducing cognitive effort than on improving decision accuracy (Einhorn, and Hogarth, 1978; Kleinmuntz, and Shchkade, 1993). The online vendors have virtually unlimited shelf space and can, therefore, offer a very large number of products to their customers. As a result, a potentially vast amount of information about market offerings is available to consumers. Searching through a marketplace composed of many such retailers would require consumers wishing to make well-informed decisions to expend a great deal of effort.

The ACE model deals with those situations where people are motivated to reason

and are capable of reasoning about the alternatives presented in a persuasive message. It proposes that people tend to use three criteria to determine whether or not they should respond favorably to the arguments of a persuader: appropriateness, consistency, and effectiveness. In the context of this study, appropriateness refers to the extent to which the recommendation is approved by important others or whether it is based on rules, norms, etc. Consistency refers to the extent to which the products recommended by a recommender system are what the individual customers or other like-minded customers would own or purchase. Effectiveness refers to the extent to which a product recommended by a recommender system leads to desired ends of the customer (Reardon, 1991).

The ACE model is a framework providing insight into how messages might be developed to evoke liking/interest and comprehension through simple association, inferences, or heuristics in reasoning about alternatives (Reardon, 1991). These three criteria can be used by the persuaders to encourage one form of behavior over another by formulating strategies to show how the persuader-recommended behavior favorably responds to those considerations.

Product Complexity

In an online shopping context, the perceived complexity of a product refers to the quantity and quality of information associated with the number of product attributes and the number of alternatives available.

Various studies have investigated the impact of product complexity on decision-

making. Bettman et al. (1998) argued that the consumer decision making processes are related to the complexity of the product in a product category. He suggested that as the complexity of the product increases, consumers are likely to resort to simpler heuristics and selective information processing, often reducing decision effectiveness. Keller and Staelin (1987) showed that as the number of attributes and alternatives increase, the decision effectiveness is reduced. Payne et al. (1993) suggested that an increased number of alternatives may result in greater cognitive load and create biases in consumer decision-making processes.

The product complexity can cause subjective information overload experience in the consumers. They would have difficulties in identifying the relevant information (Jacob, 1977), become highly selective and ignores a large amount of information (Bawden, 2001; Herbig, and Kramer, 1994), need more time to reach a decision (Jacob, 1984), and finally do not reach a decision of adequate accuracy (Malhotra, 1982). Intelligent information management systems such as recommender systems can prioritize information (Bawden, 2001; Meyer, 1998) and provide quality filter (Edmunds, and Morris, 2000; Grise, and Guallupe, 1999/2000) so that a large set of options can be reduced to a manageable size (Cook, 1993).

The recommender system organizes the product information in such a manner that the consumer is able to focus on those attributes that are most likely to maximize utility. Therefore, the impact of using a recommender system on decision-making is likely to be the greatest in a highly complex category where consumers often resort to decision making heuristics to manage the information overload.

Research Hypotheses

Studies of persuasiveness of recommender systems and other expert systems have focused on the factors influencing the end user confidence in the decision support provided by the system. Those factors include trust, explanation, transparency, and learning techniques of the recommender system. Wang and Benbasat (2004) found out that different types of explanation in recommender systems increase different trusting beliefs. The initial trust of customers can directly affect their intention to adopt the recommender system and indirectly enhance their perceived usefulness of the system. Explanation shows the reasoning process and can convince decision makers of the recommendations soundness (Ye *et al.*, 1995) because decision makers tend to discard recommendations that they do not fully understand if they will be held responsible for their decision (Hollnagel, 1987). Recommender systems have not been used in high-risk decision making for lack of transparency because it is difficult for the system to demonstrate how the decisions are reached (Herlocker *et al.*, 1999; Sinha *et al.*, 2002).

As has been discussed earlier, recommender systems use different information and algorithms in the reasoning process. They vary by 1) how much the consumers can and are willing to input, 2) the explanation capability of the algorithm, 3) what products are involved, and 4) the availability of the reference information (Ansari et al., 2000). Studies have compared recommender systems with human recommendation (Sinha *et al.*, 2001). In this study, the ACE model of persuasion (Reardon, 1991) is used to compare the persuasiveness of the two different recommender systems. The research model is presented in Figure 2.



Figure 4. Research Model

Appropriateness refers to the extent to which the recommendation is approved by important others or whether it is based on rules, norms, etc. Recommendations from both types of recommender systems are the result of its internal reasoning by either product expert or customer expert. The reasoning is based on pre-set rules. However, contentbased filtering recommender systems appear to provide recommendations that are higher on the appropriateness dimension because the reasoning logic of content-based recommender systems makes the recommended products more likely to be the right products than do the reasoning logics of collaborative-filtering recommender systems. Although collaborative filtering recommender systems are given ratings of a list of product to compare with other people's choice, a particular customer may wonder how similar those like-minded people are to him/her. Furthermore, the reasoning logic of content-based filtering recommender systems is more transparent and easily understood by customers.

Consistency refers to the extent to which the products recommended by a recommender system are what the individual customers or other like-minded customers would own or purchase. Consistency has been used as a measure for expert system acceptance (Morris, 1994) and marketing success (Chernatony, and Segal-Horn, 2003; Swait, and Erdem, 2002). Consistency involves both context and temporal dimensions. The recommendation should be consistent with the consumers' existing knowledge, which can be obtained by consumer rating of products or from preference of other likeminded people. Collaborative-filtering recommender systems appear to give product recommendations with higher consistency because they make recommendations based on like-minded customers' opinions, while content-based filtering do not consider the opinions of other like-minded customers.

Effectiveness refers to the extent to which a product recommended by a recommender system leads to the customers' desired ends. In this context, it could be argued that the usefulness of the product within the context of its intended use can be considered to be customers' desired end. It is expected that content-based filtering recommender systems will be perceived to be more effective than the recommendations

56

by collaborative-filtering recommender systems because the former explicitly ties the recommendations to customers' desired ends, while the latter does not explicitly ask about customers' desired ends nor include their desires in its reasoning logic. Therefore,

H₁: Types of recommender systems influence recommendation persuasiveness.

 H_{1a} : Content-based filtering recommender systems will be perceived to have a higher degree of appropriateness than the collaborative-filtering recommender systems.

 H_{1b} : Collaborative-filtering recommender systems will be perceived to have a higher degree of consistency than content-based filtering recommender systems.

 H_{1c} : Content-based filtering recommender systems will be perceived to have a higher degree of effectiveness than the collaborative-filtering recommender systems.

The benefit of using a recommender system is likely to be higher when the product complexity is greater (Swaminathan, 2003). When complexity of the product increases, consumers are likely to do selective information processing by concentrating on those attributes that are most likely to maximize utility in order to reduce the cognitive effort (Bettman et al., 1998).

H₂ Product complexity influence recommendation persuasiveness.

H_{2a}: Product complexity influence recommendation appropriateness.

H_{2b}: Product complexity influence recommendation consistency.

H_{2c}: Product complexity influence recommendation effectiveness.

To test the possible interaction between product complexity and types of recommender systems, a third group of hypotheses is formed:

H₃: Product complexity interacts with types of recommender system in affecting

recommendation persuasiveness.

H_{3a}: Product complexity interacts with types of recommender system in affecting recommendation appropriateness.

 H_{3b} : Product complexity interacts with types of recommender system in affecting recommendation consistency.

 H_{3c} : Product complexity interacts with types of recommender system in affecting recommendation effectiveness.

Based on Reardon's persuasion theory (Reardon, 1991), appropriateness, consistency, and effectiveness are three criteria that help consumers determine whether or not they should respond favorably to a persuader's argument when people are motivated to reason and are capable of reasoning about the alternatives presented in a persuasive message. Therefore, it is expected:

H₄: Recommendation appropriateness will be positively associated with the recommender systems' persuasiveness.

H₅: Recommendation consistency will be positively associated with the recommender systems' persuasiveness.

H₆: Recommendation effectiveness will be positively associated with the recommender systems' persuasiveness.

CHAPTER III

RESEARCH METHODOLOGY

The previous two chapters introduced the topic of the research, the review of the literature, and the hypotheses. This chapter covers the methodology used in this study. It includes the pilot study, research design explaining selection of research subjects, selection of products for experiment, selection of websites for recommender systems, the research instruments, the experiment procedure, and data analysis and statistical procedures. Much emphasis is given to the process of the research method. Based on the information gathered from the pilot study, a modified questionnaire for collecting data will be discussed.

Pilot Study

To ensure understanding of the experiment procedure instruction and survey questions by the research subjects, a convenience sample of 12 undergraduate students in MIS or business was used for a pilot test. The purpose of the pilot test is to confirm that all instructions, variables and their measurement scales are appropriate, correct, and understandable for the respondents. The feedback forms included open-ended questions to maximize the breadth of feedback and were returned anonymously. Modification of the experiment procedure instructions and survey questions was made based on the comments and recommendations.
Research Subjects

The experiments are aimed at online customers. After pilot testing the data collection instrument, a selected student sample from a junior-level Business Information System course at Mississippi State University was used for the experiment.

The adequacy of students as surrogates for non-students is a controversial issue in behavior research. Numerous studies of surrogacy have been conducted in different contexts, with mixed results. Some research studies question the external validity of results on the basis that students fail to represent non-students (Copeland, J., and Strawser, 1973; Gordon, Slade, and Schmitt, 1986; Miller, 1966; Sears, 1986). Stafford (1998) and Yavas (1994) found students to be poor surrogates of real-world consumers, since typical students often did not use the products being tested. Robinson *et al.* (1991) suggested that students and entrepreneurs differ on a variety of characteristics that are dynamic across time and situations. Others, however, argue that it is the subject's interpretation of the task and meanings they attribute that affect the external validity more than the subject type (Berkowitz, and Donnerstein, 1982). Some have found students to be adequate surrogates for real-world decision makers. For example, simulation studies that examined marketing and advertising issues have revealed no difference between managers and students (Kuehn, Khandekar, and Scott, 1996; Waters, and Collins, 1984). In an inventory management context, Mowen and Mowen (1986) found that both students and managers exhibited the same pattern of decision bias due to the way the information was framed.

Findings from different decision contexts suggest that the adequacy of student surrogates depends on the type of decision being investigated, perhaps because different decisions require different types of experience and expertise (Chang, and Ho, 2004). Lock (1986) indicates that findings of laboratory experiments can be generalized to field settings, despite the differences in subjects, tasks, and settings. He suggests that the available evidence suggest that experimental realism has a better chance of generalizing the results to other situations, settings, or people than the superficial similarity of representativeness. Therefore, even though the student sample is not representative of the overall population, and their responses may not be generalizable, students should provide adequate opinions for this exploratory study, particularly as they represent substantial online purchasing power and generally have some experience with web surfing.

Selection of Products for Experiments

Two products of different product complexity were used as the test products in this study. A variety of products have been used to study recommender systems. They include the backpacking tent and compact stereo (Haubl *et al.*, 2000), digital camera (Wang *et al.*, 2004), movie (Ansari et al., 2000; Herlocker et al., 2000), CD writer and golf club (Yoon, and Lee, 2004), tent and toothbrush (Swaminathan, 2003), and books (Sinha *et al.*, 2001). It has been suggested that when using recommender systems for product searches, collaborative filtering is more specialized than content filtering because it works based on perceived quality and tastes of people rather than objective properties (Shardanand *et al.*, 1995). Thus, recommender systems using collaborative filtering are more suited to experience goods such as books, CDs, and movies because subjective judgment that acts as the differentiator in these cases. Most of the previous studies about collaborative filtering used CDs, books and movies because of the limitation of collaborating algorithms used. Amazon.com is the company that introduced collaborative filtering to a wider number of items and people in 1998. The company's goal is to offer "something for everybody" using the online recommender system.

In order to select two products that rated high and low on product complexity, a pretest with the 14 product categories from the website using the content-based filtering recommender system (www.activebuyersguide.com) was evaluated in a focus group of 15 participants to find the two products. The website is chosen because the collaborative filtering site has much broader product categories that cover all the categories in this site. The 15 participants in the focus group were similar to the target sample. They were asked to brainstorm for products from the list that are relevant to the target population, have broad market appeal, and could be bought online. The chosen number of products was tested in a group of 30 using the 5-item, 5-point semantic differential scale for product complexity (McCabe, 1987). The resultant products with highest complexity (notebook computer) and lowest complexity (DVD player) were used in the experiment for persuasiveness study.

Selection of Websites for Recommender Systems

In order to eliminate potential confusion from complicated shopping sites and to simplify the experiment related to different recommender systems, two websites were

selected. The website for collaborative-filtering recommender system is www.amazon.com. The one for content filtering recommender system is www.activebuyersguide.com.

Although there were several commercial websites using collaborative-filtering recommender systems (i.e. www.firefly.com, www.shopping.com, and duuni.talentum.com), many of them have closed down. Amazon.com is the most successful and comprehensive one that still exits and has been used by many in collaborative filtering recommender systems research (Komiak *et al.*, 2004; Schafer *et al.*, 1999b; Sinha *et al.*, 2001; Sinha *et al.*, 2002). The online recommender systems can provide product recommendations based on a consumer's past purchase if it is a return customer. For new customers, they are first asked to rate a list of products they own or like, then the system compares the consumer's product ratings with those of other consumers. After identifying the consumer's nearest neighbors, the system recommends products that neighbors have rated highly but which the shopper may not yet have rated.

The active sales assistant site (www.activebuyersguide.com) uses content filtering recommender systems that allow consumer input of multiple product attributes (i.e. price, brand, specific functionalities, warranty, and accessories) to locate the merchant that satisfies their needs. Like most real-world content-based filtering recommender systems, it only considers a subset of all the relevant attributes in a product category. The reasons for this selectivity as stated by Haubl and Trifts (2000) include (1) the large number of attributes that exist in many product categories, (2) the substantial amount of data about a

consumer that would be required to develop an accurate understanding of the consumer's subjective preferences, (3) an inclination to use only those attributes that are common to most or all available products, (4) a tendency to include only attributes that are quantitative in nature, and (5) a strategy to emphasize or de-emphasize specific attributes.

Most of the prior studies used self-built recommender systems instead of existing commercial applications to control the initial trust or website credibility factors. There are several reasons for using two existing websites in this study.

First, prior studies do not make comparisons of different recommender systems. Most of them focus on the impact on consumer decision making with vs. without the recommender system (Desharnais *et al.*, 2002; Haubl *et al.*, 2000). The recommender systems in real world applications are all embedded in websites. One way to get around this problem is to consider the whole website as the recommender system itself.

Second, some comparison studies did use existing commercial websites with recommender systems. Sinha and Swareingen (2001) compared social recommendation from friends to online recommender systems for movie and music search. They also used five existing music websites to compare the role of transparency in recommendations provided by different online recommender systems (Sinha *et al.*, 2002).

Finally, students were asked to use different email addresses for identification each time when searching the products using the collaborative filtering recommender system by following the recommendation wizard. Cookies in the computers was disabled. Therefore, every login is registered as a new customer. This eliminated the impact on system recommendation by existing consumer buying/browsing history during the experiment (Sinha *et al.*, 2001). As will be talked about in the experiment procedure section, within-subject design and counterbalanced question orders were used to alleviate the impact from the participants.

Research Design

The research design in this dissertation was a 2x2 within-subject factorial design. The manipulated factors are: recommender systems (collaborative filtering, content-based filtering) and product complexity (high, low). Product involvement was used as a covariate. Participants were asked to purchase two different products using each recommender system. That is, each participant tried the four experiment conditions of the 2x 2 within-subject sub-design. The total number of questionnaires was equally divided into 4 parts, each part with a different combination of product and recommender system sequence, and was administered to four groups of students. This necessitated a counterbalancing of products and recommender systems usage order. The within-subject design controls for participant difference.

As a result of this experimental design, the influence of recommender systems and product complexity could be measured reliably. All other factors related to the consumer, such as emotional disposition, cognitive complexity, expectations, schemata, and expressed preferences were controlled by the within-subject design. Since all these individual characteristics should not vary in different situations, no hypotheses for them were formulated. The product involvement was used as a covariate to control an important kind of extraneous variance.

The within-subject factorial design is also called a repeated measure design. It is used when the characteristics of the participants are correlated with the dependent variable and will influence the result (Cozby, 1989). In a within-subject design, all participants receive all levels of independent variable treatment.

An obvious advantage of within-subjects design is that fewer subjects are needed, because each subject participates in all conditions. This design is preferred when subjects are scarce or when it is costly to use subjects. An additional advantage of within-subject design is that they are extremely sensitive to finding differences between groups. Because subjects in the various groups are identical in every respect (they are the same people), error variability due to subject differences is minimized (Cozby, 1989). By using counterbalancing, the unexplained variability in the result can be more readily identified, which results in a more sensitive statistical test. It is more likely to detect an effect of the independent variable if a within-subject design is used.

Two major concerns for within-subject design are the carryover effects and demand on participants. Participants need to be tested on all the treatment conditions. Therefore, fatigue becomes a greater concern. They may quickly figure out the true purpose of the experiment and behave differently than they would if they were unaware of the hypothesis. In this study, an open-ended question was used at the end of the questionnaire as a manipulation check to assess the students' awareness of the experiment purpose.

When exposed to the experiment stimulus more than once, the participants can learn how to perform the task, get tired, and change responsiveness. The learning or habituation caused can lead to a carryover effect. The carryover effect is also referred to as anchoring (Tversky, and Kahneman, 1974). When subjects are asked to give an estimate of a value for which they have little or no reference values, the first estimate will influence subsequent estimates. The tendency is to make a "reasonable adjustment" from the previous value(s). If the initial value is low, subsequent estimates tend to be at the lower end of the range. On the other hand, if the initial estimate is high, the subsequent estimates tend also to be at the higher end of the scale. The effect of basing repeated estimates on earlier figures has been described by Northcarft and Neale (1987), who asked the participants in their study to estimate real-estate values after hearing manipulated listing prices. The only influence on participant estimates was the magnitude of the listing price manipulation, whereas expertise and experience did not influence the participants estimates. Another example is the study of Block and Harper (1991), where point estimates could be influenced by being told the point estimate of another participant in the study.

The carryover effect can be minimized by counterbalancing. By arranging the various treatments in different orders for different participants, each treatment occurs in each time period of the experiment. In the end, everyone experiences the same treatments, just in different orders. If a previous measurement serves as an anchor and a "reasonable adjustment" is made, any deviation of the next measurement must be in the

direction towards the previous measurement. In a series of measurements with the factor alternating, this serves to decrease the actual difference if any anchoring takes place. Statistically significant differences between groups depend on large enough differences of the means of the groups, given the sample size and the distribution within the groups. Anchoring can potentially decrease the difference between means. It does not influence the sample size and distribution in the group, since the group size is fixed and errors are assumed to be normally distributed. If the difference of means is statistically significant, and anchoring has had an effect, the real difference is even larger, and therefore always statistically significant.

Covariate

Product involvement was used to control for extraneous variation in the dependent variables. Covariate that is uncorrelated with the other independent variables but explains significant variance in the dependent variables helps remove predictable variance from the error term, thus increasing the power of the analysis (Tabachnick, and Fidell, 1983).

According to the advertising and marketing literature, product involvement is a factor that may influence consumers' cognitive and behavioral response to online advertising. It can affect consumers information search style, process motivation, and understanding of the message (Ahn *et al.*, 2002; Cho *et al.*, 1999; Karson *et al.*, 2001; Laczniak *et al.*, 1999; Yoo *et al.*, 2001). It also varies for each customer (Koufaris et al., 2001-2002). It is likely that product involvement will have an impact on the consumers'

perception of the persuasiveness of the recommendation given by online recommender systems. Because the subject factors were controlled using the repeated measures design used in this study, product involvement was included as a covariate to investigate its possible influence on and interaction with other dependent variables.

Research Instrument

To ensure that the questions comprising the instrument were formally reviewed for content, clarity, and sequencing, two different groups of people performed a general review of the instrument. First, the questionnaire was reviewed for wording clarity and content validity by business faculty members and colleagues at Mississippi State University. Items were evaluated for ambiguity, construction faults, sequencing, and flow. Modifications were made to the instrument in accordance with their comments.

The instrument was then reviewed by the Mississippi State University Institutional Review Board for the Protection of Human Subjects. The objective of this review board is to ensure that the research being conducted by university faculty and students adheres to established research policies and procedures, and does not unnecessarily place the subjects used in the research effort at risk either mentally or physically.

Finally, the instrument was pre-tested using a convenience sample of 12 undergraduate students at Mississippi State University as mentioned in the pilot-study description. Modifications were made to the instrument in accordance with their comments prior to the formal experiment in the computer lab. The research instrument in this study is comprised of four sections (Appendix A). The first section of the instrument is a cover letter explaining the purpose of the study. The second section contains definitions of key terms that are included in other sections of the instrument. Because definitions for these terms may vary based upon their use and reference in prior research, a baseline definition for each of these terms is necessary. These definitions are based on a synthesis of various definitions offered by other research.

The third section of the instrument includes items used to measure the four dependent variables and one covariate in this study. These are effectiveness, consistency, appropriateness, persuasiveness, and product involvement. Multi-item indicators was used to measure each. The items were measured using seven point Likert-type rating scales. Because of the lack of information systems persuasiveness studies, all of the measurement items were derived from previous marketing, psychology and information systems literature related to the individual construct. Twenty items were used to measure the four constructs included in this section. Table 3 describes the operationalization of each variable along with corresponding references.

Appropriateness was measured using five items that assess the respondent's perception of the fitness and desirability of recommendations given by the recommender system. The scale used is the logo items of the Persuasive Disclosure Inventory original to Feltham (1994). According to Feltham (1994), a logos appeal provides evidence or information about a concept from which a consumer can form beliefs. The 7-point scale

has a reliability of 0.86 in the original study and 0.79 in a second verification study. One other scale, brand name appropriateness, is developed by Allen and Janiszewski (1989) who studied a subject's attitudinal evaluation of a specific brand name in constructing advertisement. In psychology, concern for the appropriateness scale is a 20-item measure of tendencies to conform to group conformity pressures. It has been used to investigate individual differences in susceptibility to peer pressure (Johnson, 1989), social anxiety (Lennox, and Wolfe, 1984), and ad context (Celuch, Slama, and Schaffenacker, 1997). This scale is not chosen for the study because the large number of items can make the data collection sessions too long and could contribute to respondent fatigue. The perception of the appropriateness of a recommendation given by an online recommender system is more like a personal evaluation of the marketing tool used by an online store. It is an individual decision with very little pressure from peers or society. The logos scale developed by Felthem was used because the brand name appropriateness is too specific for this study of recommendation appropriateness.

Consistency was measured using a scale developed by Morris (1994). The original scale includes functionality consistency and recommendation consistency, which are used to compare recommendations generated by different databases with the user's existing knowledge. Consistency has been studied heavily in marketing studies. Swait and Erdem (2002) suggest that sales promotion mix consistency can improve consumer brand evaluation. Burton et al. (1993) studied the effect of information consistency in advertisement on consumer reference-price predictions. Information consistency (i.e., all

neutral information versus one piece of positive information about the product alternative) was manipulated by Meloy (1997) to assess the role of positive affect and mood maintenance on pre-decisional distortion of evaluations of product information in preferential choice. Jain (2003) used a two-item 9-point Likert-scale to measure preference consistency and found that people are less resistant to changing preference when processing preference-inconsistent information. The items in this scale measure generalized consistency perceived by respondents. This scale can even be used for abstract concepts, but not for a specific system recommendation because it might confuse the respondents due to its ambiguity. The final scale items used were reworded according to the Purdue Consistency Testing Questionnaire (Ozok, 2000) which measures the concept consistency for information and knowledge as a subdimmension of webpage evaluation.

Effectiveness, the perceived degree of recommender systems' capability to clearly understand and address the need and concerns of the respondents, was measured by six items. Although there are many scales developed for information systems effectiveness, most of them are about large information systems in the organizational setting and none of the effectiveness measurements addressed the question in an individualized style. User satisfaction and organizational change are the two most commonly used surrogates to measure effectiveness of information systems (Gatain, 1994; Iivari, and Ervasti, 1994b; Lee, and Kim, 1995; Miller, and Doyle, 1987; Yuthas, and Young, 1998). In this study, the sales presentation effectiveness scale was used. It is a six-item, five-point Liket-type scale and is intended to measure salespeople's self-evaluation of the effectiveness of a sales presentation. The recommendation from recommender systems can be considered as a sales agent for online retailers. By giving recommendations for products that fit the consumer taste or requirements, the recommender system can promote consumer inclination towards certain products, reduce consumer search effort, improve decision quality, and increase satisfaction and loyalty to the online store. The scale also emphases the fulfillment of individual needs rather than organizational or job-related goals.

Persuasiveness refers to the extent to which customers are influenced by the reasoning of recommender systems to a belief that the recommended products best fit their personal needs (Komiak *et al.*, 2004). It was measured using (Feltham, 1994) PDI scale. The original scale includes items for measuring pathos, ethos, and logos. It has been suggested that the subscales can be used as individual message facets (Feltham, 1994). In this study, only items for ethos and pathos were used because the pathos and logos have been viewed by some as different ends of a continuum that consider the message. There are 5 items for ethos and 7 items for logos. The sum scores across the items are used to reflect ethos and pathos respectively. The original within-factor item-to-total correlations were within the range of 0.63 to 0.91, with coefficient alpha estimates of 0.89 and 0.82 respectively (Feltham, 1994). A second validation study showed the average of item-to-total correlations as 0.78 and 0.79 for ethos and logos subscale while the average reliabilities were 0.83 and 0.89 respectively. The original scale items were bipolar adjective sets using a 7-place response format. To keep them consistent with the

other scales used, the items used in the experiment were in the format of a 7-point Likert scale.

Product involvement was measured using the four-item, eight-point Likert-type summated product involvement scales by Zinkhan and Locander (1988). The original scale was developed to measure the degree of involvement a consumer has with calculators. The scale showed a reliability of 0.9 and 0.873 as reported by Zinkhan, Locander, and Leigh (1988) and Zinkhan and Locander (Zinkhan, and Leigh, 1986), respectively. Both studies found this measure multi-dimensional. Zinkhan, Locander, and Leigh (1986) used it to examine the dimensionality of several predictors of ad recall and recognition measures. Two dimensions were found, one more affective and the other more cognitive. Zinkhan and Locander (1988) used the technique to investigate four advertising recall measures and found that two dimensions were actually being measured: favorable recall of ad features and brand name recall.

 Table 3. Operationalization of Research Variables

Variable	Operationalization	Sources	Cronbach's
	1		Coefficient
			Alpha
· · ·		F 1/1 1004	0.00
Appropriateness	Mean of 5 items	Feltham, 1994	0.89
Consistency	Mean of 5 items	Morris, 1994	n/a*
Effectiveness	Mean of 6 items	Behrman &	0.8
		Perreault, 1984	
Persuasiveness	Mean of 12 items	Feltham, 1994	0.83/0.79
		,,	
Product Involvement	Mean of 4 items	Zinkhan and	0.90
		Locander, 1988	
Product Complexity	Mean of 5 items	McCabe, 1987	0.80
- F 5		.,	

Note: * indicates that the coefficient alpha value was not calculated on that study.

The fourth section of the instrument contained general information about the respondents: their age, gender, major, level of computer usage experience, level of Internet usage experience, level of online recommender system usage experience, ethnicity, and whether the respondents would like to have a copy of the major findings from the study.

All scales were assessed for face validity, construct validity, and reliability. Details were described in data analysis and statistical procedures section.

Experiment Process

The experiment was conducted in the Seal computer lab located in the College of

Business and Industry at Mississippi State University. Seal contains 30 workstations connected through the Netware local area network. Each station is equipped with a Pentium microprocessor and has multimedia capacities.

The experiment was divided into four sessions with 30 participants each. To eliminate the order effect in repeated measures design, the questionnaires were randomized by latin square design by using each of the 4 orders of the four treatments (recommender system and product complexity combinations) for each of the four experiment sessions. Cozby (1989) suggested a minimum of about 30 participants per cell in repeated measures design. Therefore, a total of 120 usable questionnaires will be the minimum needed.

Each experiment session lasted about 40 minutes. The researcher asked the participants to sign the informed consent form and explained the purpose of the experiment in general terms. Students were asked to search for two products using two different online recommender systems. The detailed instructions for the experiment are available in Appendix A. Each student had approximately seven minutes to visit each online recommender system. When each web site visit was completed, the students were asked to answer the persuasiveness questions.

Repeated measures cannot be evaluated in a random design, because the assumption of independent errors is not met. Observations made by the same participant are not independent. However, the use of a randomized complete block design is appropriate, with participants as the blocking factor. Other factors pertaining to the consumer are included in participants as the blocking factor. The measurement model is:

$$\mathbf{Y}_{ijk} = \boldsymbol{\mu} + \boldsymbol{\alpha}_i + \boldsymbol{\beta}_j + (\boldsymbol{\alpha} \boldsymbol{\beta})_{ij} + \boldsymbol{\delta}_k + \boldsymbol{\varepsilon}_{ijk},$$

where

 α_i is a fixed factor for recommender system ($\Sigma \alpha_i = 0$)

 β_i is a fixed factor for product complexity ($\Sigma \beta_i = 0$)

 α β_{ij} is interaction between recommender system and product complexity

 $(\Sigma \alpha \beta_{ij} = 0)$

 δ_k is a random factor for participant as block ($\delta_k \sim N(0,\sigma^2)$, iid)

note: iid = independent and identically distributed

N (0, σ^2) = Normal function with mean μ and variance σ^2

Data Analysis and Statistical Procedure

Face validity, construct validity and reliability were tested first. Face validity involves the systematic examination of the content of the instrument to determine whether the instrument provides adequate coverage of the problems or topics included in the study (Emory, and Cooper, 1991). An instrument is said to have a high level of face validity if it contains a representative sample of the universe of subject matter of interest. The face validity assessment procedure is a subjective process. Anastasi (1968) and Churchill (1979) suggest consulting experts who are considered knowledgeable on the research topic. One professor in marketing and two professors in MIS were asked to review the instrument.

Reliability is defined as the accuracy or precision of the research instrument and

is calculated as a proportion of the true variance to the total variance yielded by the measuring instrument (Kerlinger, 1986). Reliability can also be defined as the degree to which measures are free from error and therefore yield consistent results (Peter, 1979). One aspect of reliability is internal consistency which is an indicator of the homogeneity of a measuring scale (Cronbach, 1951). One criteria that has been consistently used to assess the reliability of multi-item measurement scales is Cronbach's (1951) coefficient alpha. Nunnally (1978) suggested that a set of items with a coefficient alpha greater than or equal to 0.7 is considered to be internally consistent. However, reliability is a necessary, but not sufficient condition, for construct validity. Thus, those aspects of construct validity were examined.

Construct validity pertains to the overall degree of correspondence between the constructs and the measures used to represent the construct (Peter, 1981). One part of construct validity is unidimensionality of the sets of items used to measure a given construct. Unidimensionality is whether items measuring a construct measure only that construct (Walizer, and Wienir, 1978). One method used to assess the unidimensionality of items is exploratory factor analysis (Harmon, 1976).

Principle component factor analysis was used to assess each of the multi-item scales in the questionnaire. Dimensionality of each factor was assessed by examining the factor loadings. Items with factor loadings of greater than .50 on the factor with which they are hypothesized to load were considered adequate indicators of that factor.

Two additional aspects of construct validity are convergent and discriminant

validity (Nunnally, 1978). Convergent validity is the extent to which a measure correlates highly with other methods designed to measure the same construct (Churchill, 1979). Principle component factor analysis was used to assess each factor for convergent validity. Factors with eigenvalues greater than 1.00 and factor loadings greater than .30 were considered to have adequate convergent validity (Caller, and Carmines, 1980).

Discriminant validity is determined by demonstrating that a measure does not highly correlate with another measure from which it should differ (Campbell, and Fiske, 1959). An examination of the cross-loadings of items on multiple factors was used to assess how well items discriminant between factors.

MANCOVA was used to evaluate hypotheses 1 to 3. Assumptions of this type of ANOVA include independence of the treatments (treatments do not influence each other), independence of the subjects, and homogeneity of variance. Within-subject measurement carries the risk of unequal variances between the subjects. Independence of the treatments was pursued by a deviation from complete randomization within the participant block, as will be explained in the next chapter. During the experiment, the researcher will request participants to not discuss anything about the study with others while the experiment is in progress. The relationship between the three ACE model constructs (appropriateness, consistency, and effectiveness) and persuasiveness as stated in hypotheses 4a, 4b, and 4c was examined using ANCOVA.

In conclusion, the research hypotheses were tested in an experimental design setting. Analysis was performed with MANCOVA for a repeated measures design for hypotheses 1-3, with recommender systems and product complexity as main effects and product involvement as a covariate. Hypotheses 4a, 4b, 4c were tested using ANCOVA. The next chapter describes the results of the data analysis.

CHAPTER IV

DATA ANALYSIS

The research objectives of this study were to examine whether there are differences in persuasiveness of recommendations generated by different recommender systems, and what are the possible factors causing the differences. Persuasiveness was also compared for products of different complexity. As an important component of persuasion process in advertising, product involvement's influence on persuasiveness was examined to establish the relevance of the findings.

These research objectives were investigated in an experimental setting, where participants were exposed to all experimental conditions in the study. By separating the variance within subjects (differences between experimental conditions) from the variance between subjects (differences between participants), statistically significant differences between mean scores for the treatments could be obtained more accurately. This chapter presents the results of data analysis. Characteristics of the sample were described followed by a discussion of data analysis such as the manipulation checks, exploratory factor analysis, MANCOVA, and evaluation of the research hypotheses.

Sample Characteristics

Participants for the study were recruited from a Business Information Systems undergraduate class at Mississippi State University. The sample consisted of one hundred and fifty-seven undergraduate students and seven were excluded after the manipulation check. Boudreau et al. (2001) advocate the use of manipulation checks in experiments, to measure the extent to which treatments have been correctly perceived by the subjects. This allowed for exclusion of results from participants who had figured out the experiment condition and the purpose of the study. In this study, the manipulation checks were conducted at the end of data collection by asking students an open-ended question about the purpose of the study. Because of the repeated measures design of this study, the students had to answer the same set of questions for four different product searches. This may produce a learning effect and the answers to the early questions can influence their perception in the later searches so that the independence of the answers will be jeopardized. Seven students indicated that the study was to compare two different recommender systems, and were eliminated from data analysis. This reduced the sample size to one hundred and fifty.

The majority of the students were between eighteen and twenty-one (90.4%) years of age, with 91 males (58%) and 66 females (42%). Because the course is a lower level required course for business majors, most of the participants were non-senior students (freshman 29.3%, sophomore 43.3%, junior 24.8%). The dominant ethnicities of the students are white (81.5%) and African American (15.3%). Students were familiar with

personal computers (mean of PC usage experience of 5-10 years) so the lack of experience with computer should not be considered as a problem. Although the students showed moderate familiarity with online shopping (mean of 4.01 on the 7 point Likertscale), they were less experienced in using online recommender systems (mean=2.52 on the 7 point Likert-scale). These characteristics were compared with those before the manipulation check as summarized in Table 4.

		N=150**	N=157**				
Age	<18	2 (1.3%)	2 (1.2%)				
	18-21	137 (91.3%)	142 (90.4%)				
	>21	11 (7.4%)	13 (8.4%)				
Gender	Male	88 (58.7%)	91 (58%)				
	Female	62 (41.3%)	66 (42%)				
Classification	Freshman	46 (30.7%)	46 (29.3%)				
	Sophomore	66 (44%)	68 (43.3%)				
	Junior	35 (23.3%)	39 (24.8%)				
	Senior	3 (2%)	4 (2.5%)				
Ethnicity	White	123 (82%)	128 (81.5%)				
	African American	22 (14.7%)	24 (15.3%)				
	Other	5 (3.3%)	5 (3.2%)				
PC usage experience	X= 5-10 yrs,	σ=1.8	X= 5-10 yrs, $\sigma = 1.8$				
Online shopping	X=4.05, σ	=0.75	X= 4.01, σ =0.7				
experience*							
Recommender Systems	$X = 2.57, \sigma = 1.74$ $X = 2.52, \sigma = 1.64$						
experience*							
* Online shopping experi-	* Online shopping experience and recommender systems experience were						
measured on a 7 point Likert-scale (1=not often, 7=very often).							
** Sample size N was 157 before the manipulation check and 150 after the							
manipulation check.							

Table 4. Sample Demographics

Data Analysis

Scree Plot and Exploratory Factor Analysis

A pilot study of 30 students was conducted to test the measurement items used in the study. An exploratory factor analysis was used to assess the unidimensionality of the multi-item scales. The unidimensionality of a set of items used to measure a given construct is a necessary, but not sufficient, condition for construct validity. Construct validity was also assessed by examining the internal consistency, and convergent and discriminant validity of each construct. Factor analysis was used to assess unidimensionality and Cronbach's coefficient alpha was used to assess internal consistency.

A principle component factor analysis was performed using the thirty-two items to measure appropriateness, consistency, effectiveness, and product involvement. The criteria used to determine the number of factors to extract was an eigenvalue that was greater than or equal to one. The result indicated that five factors had eigenvalues exceeding 1.00. This was the number of factors used in this study.

Label	Consistency	Effectiveness	Appropriateness	Persuasiveness	Involvement
cnst1	0.843				
cnst2	0.770				
cnst3	0.740				
cnst4	0.735				
effec1		0.936			
effec2		0.971			
effec3			0.782		
effec4		0.731			
effec5		0.850			
effec6		0.939			
appr1			0.764		
appr2		0.775			
appr3		0.722			
appr4			0.895		
appr5			0.835		
persu1				0.969	
persu2				0.949	
persu3				0.943	
persu4				0.863	
persu5			0.587*	0.541*	
persu6				0.783	
persu7				0.731	
persu8				0.676	
persu9				0.818	
persu10				0.754	
persul1			0.756		
persu12				0.771	
invol1					0.883
invol2					0.837
invol3					0.776
invol4	0.862				

Table 5. Exploratory Factor Analysis

Note: * indicates items containing cross loadings.

Dimensionality of each of the factors was assessed by examining the factor loadings. Items with factor loadings greater than .50 on the factor with which they are supposed to load were considered adequate indicators of that factor (Hair, Anderson, Tatham, and Black, 1995).

The evaluation of dimensionality of items yielded both expected and unexpected results (Table 5). The items measuring consistency loaded only on one factor and all loadings exceeded .70. Five out of the six items measuring effectiveness loaded on one factor while the other loaded on appropriateness at .782, indicating it does not represent a single factor. Therefore, it was eliminated from the scale. The items that were supposed to measure appropriateness loaded on two factors instead of one. Two of the five items had significant loadings on effectiveness. The other three items loaded on one factor with loadings of .764, .895, and .835 respectively. The two items that loaded on effectiveness were eliminated from the scale.

Among the twelve items measuring persuasiveness, item five cross loaded on persuasiveness and appropriateness and item eleven loaded significantly on effectiveness. All other items loaded on one factor with significant loadings. Items five and eleven were eliminated from the scale. Three of the four items for product involvement loaded on one factor while item four loaded significantly on consistency. Item four was eliminated from the scale. A second factor analysis was performed with the remaining 27 items, as a result, item 3 in effectiveness (effect 3), items two and three in appropriateness (appr 2, appr3), item five in persuasiveness (persu5), and item four in product involvement (invol4) were eliminated. A review of factor loadings from the second analysis is presented in Table 6.

Item	Consistency	Effectiveness	Appropriateness	Persuasiveness	Involvement
cnst1	0.856				
cnst2	0.750				
cnst3	0.766				
cnst4	0.747				
effec1		0.928			
effec2		0.961			
effec4		0.729			
effec5		0.840			
effec6		0.929			
appr1			0.725		
appr4			0.883		
appr5			0.861		
persu1				0.953	
persu2				0.929	
persu3				0.931	
persu4				0.861	
persu6				0.803	
persu7				0.758	
persu8				0.718	
persu9				0.847	
persu10				0.779	
persu12				0.803	
invol1					0.920
invol2					0.908
invol3					0.854

Table 6. Factor Analysis After Items Elimination

Internal Consistency

Reliability had been defined as the "degree to which measures are free from error and therefore yield consistent results" (Peter, 1979). One aspect of reliability is internal consistency, which is an indicator of the level of homogeneity of a measuring scale (Cronbach, 1951). One criterion that has been widely used to assess the reliability of a multi-item measurement scale is Cronbach's coefficient alpha. This statistical technique was used to assess the internal consistency of the model constructs. All five constructs (consistency, appropriateness, effectiveness, persuasiveness, and product involvement) had coefficient alpha values exceeding .7 (see Table 7). Nunnally suggested that a set of items with a coefficient alpha greater than .7 is considered internally consistent.

Constructs	Cronbach's Alpha
Consistency	0.910
Effectiveness	0.941
Appropriateness	0.907
Persuasiveness	0.947
Involvement	0.910

Table 7. Cronbach's Coefficient Alpha Values for Research Constructs

As discussed before, two other aspects of construct validity, convergent and discriminant validity were also assessed. The broader meaning of convergent validity has to do with the convergence of related scales and instruments. Convergent validity in this sense exists when the research proposed scale or measure of a given construct correlates with measures of the same construct using instruments proposed by other researchers. Convergent validity can also refer to the principle that the indicators of a given construct should be at least moderately correlated among themselves (Carmines, and Zeller, 1979). The result from the final factor analysis (see Table 6) indicated that all of the remaining factors had factor loadings greater than 0.5 on the factor they were supposed to load. These findings provided support for the convergent validity of the scales.

Discriminant validity refers to the principle that the indicators for different

constructs should not be so highly correlated as to lead one to conclude that they measure the same thing (Campbell et al., 1959). An examination of cross-loading of items on multiple factors provided evidence about whether items discriminate between constructs. The results from the final factor analysis (see Table 6) showed that none of the remaining items had cross-loadings on more than one factor. Therefore, the constructs exhibited adequate discriminant validity.

MANCOVA

After elimination of cross-loading items and subjects that failed the manipulation check, the sum score for each experimental condition and participant was calculated. These results were analyzed with MANCOVA in a 2 x 2 factorial design with participants as block and product involvement as a covariate for persuasiveness. The results showed that the blocking for participants was effective (p<0.001). In other words, using the block design was effective in separating between-subject variance from within-subject variance. For persuasiveness, there was no interaction between recommender systems and product complexity (p=0.22), but significant difference between the two recommender systems (p<0.001). Product complexity had no statistically significant impact on recommendation persuasiveness (p=0.58). The effect of product involvement was also insignificant (p=0.99). The results of this analysis can be seen in Table 8. Similar results can be found for consistency, appropriateness, and effectiveness, and appropriateness. There was no interaction between system and product complexity and system made differences in consistency, effectiveness, and appropriateness. There was no interaction between system and product complexity

for consistency, effectiveness, and appropriateness. Product involvement did not show a significant impact on consistency, effectiveness, and appropriateness.

Source	Dependent	Type III	df	Mean	F	Pr>F	
	Variable	Sum of		Square			
		Square					
Total (corr.)	appropriateness	8596.80	599				
	consistency	19819.20	599				
	effectiveness	25869.80	599				
	persuasiveness	80343.20	599				
Model	appropriateness	4014.58	153	26.24	2.55	<0.001*	
	consistency	10623.44	153	69.43	3.37	<0.001*	
	effectiveness	11225.82	153	73.37	2.23	<0.001*	
	persuasiveness	56831.68	153	371.45	7.05	<0.001*	
Participant	appropriateness	3569.05	149	23.95	2.33	<0.001*	
-	consistency	9965.9	149	66.86	3.24	<0.001*	
	effectiveness	10200.05	149	68.46	2.08	<0.001*	
	persuasiveness	54109.23	149	363.15	6.89	<0.001*	
System	appropriateness	422.04	1	422.04	43.03	<0.001*	
	consistency	651.04	1	651.04	31.85	<0.001*	
	effectiveness	1011.40	1	1011.40	30.08	<0.001*	
	persuasiveness	885.75	1	885.75	16.8	<0.001*	
Complexity	appropriateness	0.38	1	0.38	0.04	0.85	
	consistency	4.68	1	4.68	0.23	0.63	
	effectiveness	13.20	1	13.20	0.40	0.53	
	persuasiveness	16.45	1	16.45	0.31	0.58	
System X Complexity**	appropriateness	0.38	1	0.38	0.04	0.85	
	consistency	0.04	1	0.04	0.00	0.96	
	effectiveness	0.20	1	0.20	0.01	0.94	
	persuasiveness	81.4	1	81.4	1.54	0.22	
Involvement	appropriateness	2.74	1	2.74	0.27	0.61	
	consistency	1.73	1	1.73	0.08	0.77	
	effectiveness	0.97	1	0.97	0.03	0.87	
	persuasiveness	0.014	1	0.014	0.00	0.99	
Error	appropriateness	4582.22	446	10.27			
	consistency	9195.76	446	20.63			
	effectiveness	14643.98	446	32.83			
persuasiveness 23511.52 446 52.27							
Note: model with participants as block, product involvement as covariate, and system and							
complexity as main effects.							
* significant ** interaction							

Table 8. MANCOVA of Research Model

When the model was evaluated again with only the recommender systems, and participants as blocks, both blocking and recommender systems retained their statistical significance, and product involvement remained insignificant (see Table 9). Both participants and system had p values less than 0.001 for persuasiveness, consistency, effectiveness, and appropriateness.

Source	Dependent	Type III	df	Mean	F	Pr>F	
	Variable	Sum of		Square			
		Square					
Total (corr.)	appropriateness	8596.80	599				
	consistency	19819.20	599				
	effectiveness	25869.80	599				
	persuasiveness	80343.2	599				
Model	appropriateness	4014.21	151				
	consistency	10617.22	151	70.31	3.42	<0.001*	
	effectiveness	11216.21	151	65.62	2.01	<0.001*	
	persuasiveness	56733.90	151	375.72	7.13	<0.001*	
Participant	appropriateness	3431.57	149	23.03	2.25	<0.001*	
	consistency	9229.31	149	61.94	3.02	< 0.001*	
	effectiveness	9777.57	149	65.62	2.01	< 0.001*	
	persuasiveness	54274.14	149	364.26	6.91	<0.001*	
System	appropriateness	442.20	1	442.20	43.23	<0.001*	
	consistency	651.09	1	651.09	31.70	<0.001*	
	effectiveness	1011.10	1	1011.10	30.91	<0.001*	
	persuasiveness	885.94	1	885.94	16.81	<0.001*	
Involvement	appropriateness	3.12	1	3.12	0.31	0.58	
	consistency	0.23	1	0.23	0.01	0.92	
	effectiveness	4.76	1	4.76	0.15	0.70	
	persuasiveness	271	1	2.71	0.05	0.82	
Error	appropriateness	4582.59	448	10.23			
	consistency	9201.97	448	20.54			
	effectiveness	14653.59	448	32.71			
	persuasiveness	23609.30	448	52.70			
Note: model with participants as block, product involvement as covariate, and system and							
complexity as main effects.							
* significant							

Table 9. MANCOVA of Research Model (without product complexity)

As shown in Table 10, subsequent comparison of means using the Least Significant Difference (LSD) demonstrated that the content-based recommender system showed statistically significant better appropriateness, consistency, effectiveness and persuasiveness than the collaborative filtering recommender system.

	t-Grouping	Mean*	N**	System		
Consistency	А	25.61	300	Content-based		
	В	23.53	300	Collaborative		
Effectiveness	А	24.45	300	Content-based		
	В	21.85	300	Collaborative		
Appropriateness	А	15.96	300	Content-based		
	В	14.24	300	Collaborative		
Persuasiveness	А	38.78	300	Content-based		
	В	36.35	300	Collaborative		
* denotes mean	for groups with	n differen	t letter	are significantly		
different.						
** N is number of subjects in each group.						

Table 10. Comparison of Means for Recommender Systems

As product complexity did not show a significant impact on recommendation persuasiveness and there was no significant interaction between system and product complexity, the comparison of the means for combination of recommender system and product complexity was similar to those for comparison of groups using different systems. The result can be seen in Table 11. There were no differences between the means of persuasiveness, consistency, appropriateness, and effectiveness for combination groups using the same recommender systems although the products were of different complexity.

	t-Grouping	Mean*	System	
Consistency	A	23.60	Collaborative & DVD	
		23.44	Collaborative & Notebook	
	В	25.95	Content-based & DVD	
		25.66	Content-based & Notebook	
Effectiveness	Α	26.60	Collaborative & DVD	
		26.37	Collaborative & Notebook	
	В	30.14	Content-based & DVD	
		29.73	Content-based & Notebook	
Appropriateness	Α	24.21	Collaborative & DVD	
		24.17	Collaborative & Notebook	
	В	27.07	Content-based & DVD	
		27.31	Content-based & Notebook	
Persuasiveness	Α	42.79	Collaborative & DVD	
		44.23	Collaborative & Notebook	
	В	47.05	Content-based & DVD	
		46.49	Content-based & Notebook	
* denotes mean for groups with different letters are significantly				
different.				

Table 11. Comparsion of Means for Recommender System &

Product Complexity

As stated in the previous chapter, the possibility of anchoring is a concern in a study with repeated measures design. Kahneman and Tversky (1974) compared mean responses after the anchor. In this study, two different recommender systems were alternated as the first one to be used for product searching. The anchoring was tested by comparing the means of persuasiveness for the content-based recommender system and collaborative filtering recommender system in the first and second position for each. In the case of the content-based recommender system, the means for persuasiveness in the first and in the second place did not differ significantly with a p-value of 0.925, but for

the collaborative recommender system, the p-value was 0.0005, which showed a significant difference. These values were calculated with the paired t-test because of the close to normal distribution for the groups as evidenced by the histograms.

Finally, the relationships between persuasiveness and consistency, effectiveness, and appropriateness were evaluated. All factors (subject, effectiveness, and appropriateness) except consistency contribute significantly to the variance in persuasiveness with a R-square of 0.86. The results are summarized in Table 12.

Source	df	Sum of Square	Mean Square	F value	Pr>F	
Total (corr)	599	80343.198				
Model	153	68895.13	450.30	17.54	<0.0001*	
Consistency	1	2.76	2.76	0.11	0.743	
Effectiveness	1	1134.15	1134.15	44.18	< 0.0001*	
Appropriateness	1	1389.97	139.97	54.15	<0.0001*	
Involvement	1	1.52	1.52	0.06	0.81	
Subject	149	34874.74	234.06	9.12	<0.0001*	
Error	446	11448.06	25.67			
Note: model with participants as block and product involvement as covariate.						
* significant						

 Table 12. ANCOVA of Persuasiveness

Although product involvement has been suggested as an important factor influencing response to online advertising, none of the tests in this study showed any significant impact. This may be caused by the control of the subject factor in the repeated measures design because Koufaris (2001-2002) indicated that product involvement is associated with the consumer. Therefore, the data did not provide an adequate foundation to assess the impact of product involvement on recommender system persuasiveness.

Evaluation of the Research Hypotheses

It is clear from the description of the data analysis that some but not all research hypotheses were supported. Results of the MANCOVA indicated that the different recommender systems led to different persuasiveness, but the persuasiveness was not influenced by product complexity. Furthermore, the impact of product involvement was not significant because of the extent of subject control in the repeated measures design. The first hypothesis tested whether types of recommender systems would influence recommendation persuasiveness. Analysis of the results showed that the persuasiveness of recommendation from different recommender systems differs significantly. Therefore, hypothesis H₁ is supported.

Research hypotheses H_{1a} , H_{1b} , and H_{1c} concerned the influence of types of recommender systems on appropriateness, consistency, and effectiveness. Content-based filtering recommender systems were hypothesized to have a higher degree of appropriateness and effectiveness, and collaborative filtering recommender systems were hypothesized to have a higher degree of consistency. As shown in the tables below, content-based filtering recommender systems have higher mean scores for appropriateness, effectiveness, and consistency. Therefore, H_{1a} and H_{1c} were supported and H_{1b} was not supported. It was clear that the scores for the three dependent variables
for the two different recommender systems belong in different groups (see Tables 13-15).

	t Grouping	Mean*	Ν	System
Appropriateness	А	15.96	300	Content-based
	В	14.24	300	Collaborative
* denotes mean for groups with different letters are significantly different.				
N=observations in group				

Table 13. Comparison of Mean for H_{1a}

Table 14. Comparison of Mean for H_{1b}

	t Grouping	Mean*	Ν	System
Consistency	А	25.61	300	Content-based
	В	23.53	300	Collaborative
* denotes mean for groups with different letters are significantly different.				
N=observation in group				

Table 15. Comparison of Mean for H_{1c}

	t Grouping	Mean*	Ν	System
Effectiveness	А	24.45	300	Content-based
	В	21.85	300	Collaborative
* denotes mean for groups with different letters are significantly different.				
N=observation in group				

Hypothesis H₂ is related to the influence of product complexity on persuasiveness,

effectiveness, consistency, and appropriateness. It was hypothesized that different levels

of product complexity will lead to differences in the four dependent variables. The results showed no impact of product complexity on them (see Tables 16-19). Therefore, H_2 , H_{2a} , H_{2b} , and H_{2c} were not supported. Part of the reason may lie in the relatively small differences in complexity between Notebook computer and DVD players.

Table 16. Comparison of Means for H₂

	t Grouping	Mean*	Ν	System
Persuasiveness	А	37.75	300	Notebook
	А	37.39	300	DVD player
* denotes mean for groups with same letters are not significantly different.				
N=observation in group				

Table 17. Comparison of Means for H_{2a}

	t Grouping	Mean*	Ν	System
Appropriateness	А	15.13	300	Notebook
	А	15.08	300	DVD player
* denotes mean for groups with same letters are not significantly different.				
N=observation in group				

Table 18. Comparison of Means for H_{2b}

	t Grouping	Mean*	Ν	System
Consistency	А	24.66	300	Notebook
	Α	24.48	300	DVD player
* denotes mean for groups with same letters are not significantly different.				
N=observation in group				

Table 19. Comparison of Means for H_{2c}

	t Grouping	Mean*	Ν	System
Effectiveness	А	23.30	300	Notebook
	А	23.00	300	DVD player
* denotes mean for groups with same letters are not significantly different.				
N=observation in group				

Hypothesis H3 is to test the interaction between recommender systems and product complexity to affect persuasiveness, appropriateness, consistency and effectiveness. As there is no interaction witnessed in the MANCOVA test (see Tables 8), H3, H3a, H3b, and H3c were not supported.

The next research hypotheses concerned the persuasion theory, whether higher appropriateness, consistency and effectiveness of the recommendation lead to better persuasiveness. ANCOVA showed that consistency, effectiveness, and appropriateness together explained most of the variances in persuasiveness (85.75% with product involvement and 85.23% without product involvement). While effectiveness and appropriateness were significant factors in determining recommendation persuasiveness (both p<0.0001), consistency did not show statistical significance (p=0.74 with involvement and p=0.98 without involvement). Subject was a significant factor for recommendation persuasiveness. This proved the appropriateness of using subject controls in the study.

There are several significant findings in the study. First, system recommendations

for non-experience product types are more persuasive than for collaborative filtering recommender systems. This includes higher scores for appropriateness, consistency, and effectiveness. Although the collaborative filtering system considers consumer purchasing history, it did not show better consistency in its recommendation as claimed. Product complexity was a non-significant factor in determining persuasiveness. Product involvement, which has been suggested as a possible influence factor, cannot be proved to have a significant impact on persuasiveness in this study.

In summary, the study was successful in confirming some, but not all, research hypotheses. The results of the evaluation of all hypotheses were summarized in Table 20. H_1 , H_{1a} , H_{1c} , H_4 , and H_5 are supported while the rest of the hypotheses are not supported by the data collected in the study.

Table 20. Summary of Hypothesis Testing

Hypothesis	Result			
H1: Types of recommender systems influence recommendation	Supported			
persuasiveness.				
H1a: Content-based filtering recommender systems will be perceived				
to have a higher degree of appropriateness than the collaborative				
filtering recommender systems.				
H1b: Collaborative filtering recommender systems will be perceived	Not			
to have a higher degree of consistency than content-based filtering	supported			
recommender systems.				
H1c: Content-based filtering recommender systems will be perceived	Supported			
to have a higher degree of effectiveness than the collaborative				
filtering recommender systems.				
H2: Product complexity influence recommendation persuasiveness.	Not			
	supported			
H2a: Product complexity influence recommendation appropriateness.	Not			
	supported			
H2b: Product complexity influence recommendation consistency.				
	supported			
H2c: Product complexity influence recommendation effectiveness.	Not			
	supported			
H3: Product complexity interacts with types of recommender systems				
in attecting recommendation persuasiveness.				
H3a: Product complexity interacts with types of recommender	Not			
systems in affecting recommendation appropriateness.	supported			
H3b: Product complexity interacts with types of recommender	Not			
systems in affecting recommendation consistency.	supported			
H3c: Product complexity interacts with types of recommender	Not			
systems in affecting recommendation effectiveness.				
H4: Recommendation appropriateness will be positively associated				
with the recommender system's persuasiveness.				
H5: Recommendation consistency will be positively associated with				
the recommender system's persuasiveness.	supported			
H6: Recommendation effectiveness will be positively associated with	Supported			
the recommender system's persuasiveness.				

CHAPTER V

CONCLUSION AND IMPLICATIONS

Research on online recommender systems has focused on the statistical accuracy of the algorithms driving the system, with little emphasis on interface issues and the user perspective. Some recent studies have explored the impact of transparency (Herlocker et al., 2000), explanation (Wang et al., 2004; Ye et al., 1995), trust (Wang, and Benbasat, 2005) on acceptance of recommendations given by recommender systems but none of them has compared those influences in different systems. In this study, a new concept, the persuasiveness of recommendations given by two types of recommender systems, was compared. Based on Reardon's persuasion theory, the underlying reasons for the differences were investigated.

The two types of recommender systems, the content-based and the collaborative filtering recommender systems did show significant differences in persuasiveness. The recommendation given by the content-based recommender systems are more believable, credible and trustworthy to the consumers searching for products. The better recommendation from the content-based recommender systems was also more appealing to the consumers because they provided affective and stimulating information matching the consumer needs. The content-based recommender system showed higher scores for

all three dimensions of persuasiveness introduced by Reardon. By asking customer input of product features and customer preferences, the content-based recommender system clarified the customer needs and requirements. By arranging the best-match recommendation next to the preferred specifications, the system made the reasoning process and logic transparent to the customers so that they can tell whether their requirements have been met. Therefore, they can determine whether the recommendation was appropriate and effective compared with their explicit needs. The content-based recommender system was better in revealing the ties between customer needs and product recommendation. This made the recommendations more persuasive to the customer.

Although the collaborative filtering recommender system provided a way to help product search when customers were not sure or cannot clearly express their requirements or preferences, the system scored lower on consistency than did the content-based systems. The way the system soliciting customer preference did not present the criteria or rules the system used to search for what the like-minded people want. Therefore, the customer cannot determine whether the recommended products were consistent to their taste/preference or existing knowledge. This is different from H1b. Although a contentbased recommender system did not consider information of like-minded people, it still showed better consistency in matching recommendations with customer guidance. Instead of getting their existing knowledge and taste from the products rating, the product features were considered to be a good representation of consumer preferences, which is more explicit and accurate for the kind of product selected for the current study. Although studies (Charlet, 1998a) have suggested that the collaborative filtering recommender systems are more appropriate for searching experience goods like books, CDs, and movies, amazon.com has claimed its recommender system capability in search of most products. The results from this study showed that for non-experience products, the content-based recommender systems were perceived to be more persuasive by providing more appropriate, more effective, and more consistent recommendations.

When using recommender systems to give personalized product recommendations for consumers, online retailers should make sure that the types of systems are appropriate for the product types. For non-experience products for which the product attributes can be explicitly expressed, content-based recommender systems can provide more persuasive recommendations. While the use of collaborative filtering recommender systems for nonexperience products leads to less persuasive recommendations, its repeated occurrence can have a negative impact on consumer perception of an online retailer and an online store. Consumers will look for other online stores that can provide more persuasive recommendations. Therefore, to fully realize the intended purpose of their recommender systems, online retailers should tailor the recommender systems used for different product categories so that persuasive recommendations could be obtained by consumers visiting their online store.

Product complexity did not show significant impact on recommendation persuasiveness even though it has been suggested as a factor reducing the effectiveness and accuracy of complex consumer decision-making (Keller et al., 1987). There are two

103

explanations for this outcome. First, the recommender systems have effectively reduced the complexity of the decision making for consumer product search by information filtering and priority management. The amount of alternatives and attributes associated with a product is no longer a burden to the consumer. The second reason could be the small differences of product complexity between products used in the study. Therefore, there is a wide range of products with different complexity that online retailers can use for their recommender systems. The recommender system can be a good tool to alleviate the barriers in selling complex product online.

According to Reardon's (1981) persuasion theory, appropriateness, consistency, and effectiveness of a persuasion message can determine the persuasiveness of the message when people are motivated to reason and are capable of reasoning about the alternatives presented. The result of the study showed a significant difference between the consistency of the recommendation given by the two recommender systems, and yet it was not a significant factor in determining message persuasiveness. Appropriateness and effectiveness of the persuasion message can be easily communicated to the consumer by detailed explanation mechanism and comparison between the consumer input and recommendation output.

Compared to appropriateness and effectiveness, consistency is a factor that is more difficult for consumers to evaluate. It is hard to explain explicitly and can involve different criteria. Although the collaborative filtering recommender systems are designed to connect consumer past purchasing history or preference of like-minded people with the recommended product, most of them were not able to demonstrate a logic link between the consumer input and system recommendation. Consistency can be an insignificant determinant of persuasiveness when the search product is non-experience goods and the consumer needs can be materialized into specific attributes. Online retailers should try to avoid using collaborative filtering recommender systems for personalized product recommendation for non-experience goods. When using collaborative filtering recommender systems for experience goods, a good presentation of the reasoning logic should be helpful for consumers to accept the system recommendation. The usefulness of consistency for persuasion message needs to be determined by further study.

Appropriateness and effectiveness are two criteria that can be used by online retailers to assess their online recommender systems persuasiveness. These differ from usefulness and ease of use when evaluating online recommender systems acceptance. They provide a complementary set of measurements that can help online retailers improve the effectiveness and efficiency of their online recommender systems. Therefore, appropriate recommender systems could be implemented for the suitable products categories to achiever better product targeting. For example, when a consumer asks the recommender system to search for a non-experience product, a content-based recommender system will be started asking for product attributes and specifications. If an experience product was input for searching, a collaborative filtering recommender system will be engaged for preference searching either by giving a list of products for rating or by considering the consumer's past history for clues to potential choices. Providing collaborative filtering recommender systems for recommendation of all product categories can lead to unsatisfied customers. The negative impression of an unpersuasive recommendation could affect the future utilization of the recommender system. This can lead to underutilization of the recommender systems invested by the online retailers and drive the potential customers to competitor sites.

When searching for products of different complexity, the two recommender systems did not show significant difference in persuasiveness. The use of recommender systems by online retailers can help consumers dealing with complex information. The recommender systems help the consumers overcome the difficulties in processing information associated with products of high complexity. By filtering and prioritizing product information of customer preference, the consumers were able to reach quick decisions with adequate accuracy. If the recommender systems were well-matched with the appropriate product categories, retailers can effectively increase the kinds of product that could be marketed online.

In addition to the practical implications, this study contributes to knowledge about persuasiveness in the IS community. Although persuasiveness has been a well-researched subject in communication (Block, and Keller, 1997; Pornpitakpan, 2004; Strahan, Spencer, and Zanna, 2002), psychology , and advertising in marketing (Anand, and Sternthal, 1992; Komiak et al., 2004; Lowrey, 1998), it is new in IS research. Persuasiveness provides another perspective for IS research that extends this concept from interpersonal interaction to human-computer interaction. This enriches the widely used technology acceptance models. The last decade has seen phenomenal growth and improvement in the capabilities of computer-based information systems, especially the internet based applications, yet the full utilization of the technology capabilities are limited by the human component in the chain. Therefore, it is important to introduce new concepts in understanding the human behavior so that the weakest link can be strengthened for better technology adoption and utilization.

The use of repeated measures design is another contribution of the research. With this design, the subject factor in the persuasive communication can be controlled so as to isolate the persuasion message. This is not feasible with other studies before. The successful control of the subject factor can be proved by the result about the covariate used.

Product involvement has been suggested as an influencing factor for cognitive and behavioral response to online advertising in marketing research. It can affect consumer information search style, process motivation, and understanding of the message (Ahn et al., 2002; Cho et al., 1999; Karson et al., 2001; Laczniak et al., 1999; Yoo et al., 2001). Product involvement varies for each customer (Koufaris et al., 2001-2002). The current study showed that persuasiveness as well as its three inducements (appropriateness, consistency, and effectiveness) differs significantly between subjects but product involvement did not show significant impact on them. Because of the repeated measures design, the variance caused by the difference in product involvement has been expressed in the subject factor and there is no single factor effect shown for product involvement. Therefore, this design alleviated the difficulties in isolating the multiple factors in the persuasion process and concentrated on the study of persuasion message development and presentation by different recommender systems.

Limitations and Future Research

Every study has its strengths and weaknesses, and this study is no exception. One of the weaknesses in experimental designs is limited realism to the participants. Even though students are the largest population group for online shopping, they cannot completely surrogate the actual online customer population. Participants in the main sample were primarily undergraduate students under the age of 21. Though the products used in the experiment were picked from the focus group as the most likely online purchase items for college students, there is no limit set for the prices and affordability for the two products. It can reasonably be expected that only a few had real need for the product searched which might affect their information search behavior and perception of the recommendation persuasiveness. The levels of complexity of the two products chosen are comparatively close which may contribute to the insignificant impact of product complexity on persuasiveness. In future studies, different population samples and different products can be used.

The sample was also very homogenous with respect to age and level of education, and the same study with participants significantly different in age or education could have produced different results. The students' average experience with online recommender systems is at the lower end but there are several participants who have used the recommender system very often. Since experience with recommender systems is not a factor considered in this study, their answers can skew the overall response to the survey questions. Until the study has been repeated with a significantly different sample, the generalization will remain low.

A further limitation of this study is the carryover effect. The carryover effect is an inherent problem with repeated measures design. The results of the study showed that for the collaborative filtering recommender system, the participants showed a significant difference toward recommendation persuasiveness when the system was used first and second for product search. This brought possible bias to the results due to the learning participants gained in using the systems repeatedly. Future studies could use the randomized design with identical subject groups so that different experiment treatments could be given to each subject only once to eliminate the carryover effect. In this new design, product involvement can be added as a third treatment which is not feasible in a repeated measures design.

The use of existing websites with different recommender systems also introduced limitations to the study. While the use of existing sites added more realism to the study, many non-recommender system factors can be introduced. Although the cookies were disabled and participants were asked to register as new users every time starting a new search, there were obvious differences in familiarity with the websites used according to the time students spent on the experiment tasks. Even though very few students said they have used the recommender systems, their previous experience with other functions at the websites can affect their response to the questionnaires in the current study. Future studies could try building new recommender systems with different algorithms and technologies rather than using ones that are currently available from commercial websites. This ensures that the agent would be new to all participants and the study would remain focused on the initial differences in the system.

Due to the advances in web-based technologies and the explosive amount of information available to consumers, more companies are using online recommender systems to provide personalized product information to their customers. However, because of the high risk and uncertainties associated with online environment, consumers must be convinced when doing product search or actual purchase online. According to Wang and Benbasat (2005), recommender systems are treated as "social actors" with human characteristics. Therefore, the persuasion theories in the interpersonal domain may generally apply to persuasiveness in technological artifacts. More research is needed to examine other factors in the persuasion process such as trust, source credibility, consumer self-confidence. For example, Wang and Benbasat (2003) have explored that consumer trust in the recommender system as "virtual assistants" are important in consumers' intentions to adopt online recommenders systems.

Studies suggested that consumers' social relationship with online stores could be strengthened by further interactions. Additional interaction with the online store may reduce consumers' perception of uncertainty and risk in using them (Gefen et al., 2003). This may apply to persuasiveness of recommender systems. The insignificance of consistency for persuasiveness in the current study may be caused by the omission of time factor. Additional research is needed to investigate the importance of consistency and change of persuasiveness over time.

BIBLIOGRAPHY

- Ahn, E., and Edwards, S. M. (Year). "Does Size Really Matter Brand Attitude Versus Click-through in Response to Banner Ads." Paper presented at the The 2002 Conference of the American Academy of Advertising, Auburn, AL, 2002.
- Aldag, R. J., and Power, D. J. (1986). An Empirical Assessment of Computer-assisted Decision Analysis. *Decision Sciences* 17, 572-588.
- Allen, C. T., and Janiszewski, C. A. (1989). Assessing the Role of Contingency Awareness in Attitudinal Conditioning with Implications for Advertising Research. *Journal of Marketing Research* 26, 30-43.
- Anand, P., and Sternthal, B. (1992). The Effect of Program Involvement and Ease of Message Counterarguing on Advertising Persuasiveness. *Journal of Consumer* psychology 1, 225-239.
- Anastasi, A. (1968). *Psychology Testing* (3rd ed.). MacMillan Company, London, England.
- Andreson, N. H. (1971). Integration Theory and Attitude Change. *Psychological Review* 78, 171-205.
- Ansari, A., Essegaier, S., and Kohli, R. (2000). Internet Recommendation Systems. *Journal of Marketing Research* 37, 363-375.
- Areni, C. S., Ferrell, M. E., and Wilcox, J. B. (2000). The Persuasive Impact of Reported Group Opinions on Individuals Low vs. High in Need for Cognition: Rationalization vs. Biased Elaboration? *Psychology & Marketing* 17, 855-875.
- Bandura, A. (1977). Social Learning Theory. Prentice-Hall, Englewood Cliffs, NJ.
- Barki, H., and Hartwick, J. (1994). Measuring User Participation, User Involvement, and User Attitude. *MIS Quarterly* 18, 59-82.

- Baroudi, J. J., Olson, M. H., and Ives, B. (1986). An Empirical Study of the Impact of User Involvement, and User Attitude. *Communications of the ACM* 29, 232-238.
- Bary, J. H., and Maxwell, S. E. (1985). *Multivariate Analysis of Variance*. Sage Publications, Inc., Newbury Park, CA.
- Basu, C., Hirch, H., and Cohan, W. (Year). "Recommendation as Classification: Using Social and Content-Based Information in Recommendation." Paper presented at the 1998 National Conference on Artificial Intelligence, Madison, Wisconsin, 1998.
- Bawden, D. (2001). Information Overload. Library & Information Briefings 92
- Behrman, D. N., and Perreault Jr., W. D. (1984). A Sales Role Stress Model of the Performance and Satisfaction of Industrial Salespersons. *Journal of Marketing* 48, 9-21.
- Berkowitz, L., and Donnerstein, E. (1982). External Validity is More Than Skin Deep. *American Psychologist* 37, 245-257.
- Bettman, J., Luce, M. F., and Payne, J. W. (1998). Constructive Consumer Choice Processes. *Journal of Consumer Research* 25, 187-217.
- Bettman, J. R. (1979). An Information-Processing Theory of Consumer Choice. Addison-Wesley, Glen View, IL.
- Bhargava, H., and Power, D. J. (Year). "Decision Support Systems and Web Technologies: A Status Report." Paper presented at the Americas Conference on Information Systems 2001, Boston, MA, August 3 5, 2001.
- Bierley, C., McSweeney, F. K., and Vannieuwlerk, R. (1985). Classical Conditioning of Preferences for Stimuli. *Journal of Consumer REsearch* 12, 316-323.
- Block, L. G., and Keller, P. A. (1997). Effects of Self-Efficacy and Vividness on the Persuasiveness of Health Communications. *Journal of consumer psychology* 6, 31-55.
- Block, R. A., and Harper, D. R. (1991). Overconfidence in Estimation: Testing the Anchoring-and-Adjustment Hypothesis. *Organizational Behavior and Human Decision Processes* 49, 188-207.
- Bottoms, D. (1995). Making the "Net Pay Off". Industry Week 244, 57-62.

- Boudreau, M. C., Gefen, D., and Straub, D. W. (2001). Validation in Information Systems Research. *MIS Quarterly* 25, 1-14.
- Boyd, H. W., Walker, O. C., Mullins, J., and Larreche, J.-C. (2002). *Marketing Management, A Strategic Decision-Making Approach*. McGraw-Hill/Irwin, Columbus, OH.
- Brassington, F., and Pettitt, S. (2003). *Principles of Marketing* (3rd ed.). Prentice-Hall/Finanical Times, Englewood Cliffs, NJ.
- Breese, J., Heckerman, D., and Kadie, C. (Year). "Empirical Analysis of Predictive Algorithms for Collaborative Filtering." Paper presented at the 14th Conference on Uncertainty in Artificial Intelligence, Madison, Wisconsin, 1998.
- Burton, S., Lichtenstein, D. R., and Herr, P. M. (1993). An Examination of the Effects of Information Consistency and Distinctiveness in a Reference-Price Advertisement Context. *Journal of Applied Social Psychology* 23, 2074-2093.
- Buttman, J. R., Johnson, E., and Payne, J. W. (1990). A Componential Analysis of Cognitive Effort in Choice. *Organizational Behavior and Human Decision Processes* 45, 111-139.
- Caller, R. A., and Carmines, E. G. (1980). *Measurements in Social Sciences: The Link Between Theory and Data*. Cambridge University Press, New York, NY.
- Campbell, D. T., and Fiske, D. W. (1959). Covergent and Discriminant Validation by the Multitrait-Multimethod Matrix. *Psychological Bulletin* 56, 81-105.
- Carmines, E. G., and Zeller, R. A. (1979). *Reliability and Validity Assessment*. Sage Publications, Newbury Park, CA.
- Celuch, K., Slama, M., and Schaffenacker, S. (1997). Concern for Appropriateness and Ad Context Effects. *Psychology & Marketing* 14, 19-28.
- Chaiken, S. (1980). Heuristic vs. Systematic Information Processing and Use of Sources vs. Message Cues in Persuasion. *Journal of Personality and Social Psychology* 39, 752-766.
- Chaiken, S., and Maheswaran, D. (1994). Heuristic Processing Can Bias Systematic Processing: Effects of Source Credibility, Argument Ambiguity, and Task Importance on Attitude Judgment. *Journal of Personality and Social Psychology* 66, 460-473.

- Chang, C. J., and Ho, J. L. Y. (2004). Judgment and Decision Making in Project Continuation: A Study of Students as Surrogates for Experienced Managers. *ABACUS* 40, 94-116.
- Charlet, J.-C. (1998a). Rules-Based Systems. Standford University Graduate School of Business Technology notes
- Charlet, J.-C. (1998b). Collaborative Filtering. Standford University Graduate School of Business Technology notes
- Chernatony, L. D., and Segal-Horn, S. (2003). The Criteria for Successful Services Brands. *European Journal of Marketing* 37, 1096-1118.
- Cheung, C. M. K., Zhu, L., Kwong, T., Chan, G. W. W., and Limayem, M. (Year). "Online Consumer Behavior: A Review and Agenda for Future Research." Paper presented at the 16th Bled eCommerce Conference, Bled, June, 2003.
- Chiasson, M., and Dexter, A. S. (2001). System Development Conflict During the Use of an Information System's Prototyping Method of Action Research: Implications for Practice and Research. *Information Technology & People* 14, 91-108.
- Cho, C. H., and Leckenby, J. (1999). How Advertising Works on the WWW: Modified Elaboration Likelihood Model. *Journla of Current Issues and Research in Advertising* 21, 33-50.
- Churchill, G. A. (1979). A Paradigm for Developing Better Measures of Marketing Constructs. *Journal of Marketing Research* 16, 64-73.
- Clancy, H. (1995). Internet, Web: The Marketing Tools of the 21th Century. *Computer Reseller News* 631, 43, 45.
- Cook, G. J. (1993). An Empirical Investigation of Information Search Strategies with Implications for Decision Support System Design. *Decision Sciences* 24, 683-699.
- Copeland, R. M., J., F. A., and Strawser, R. H. (1973). Students as Subjects in Behavioral Research. *The Accounting Review* 191-207.
- Cozby, P. C. (1989). *Methods in Behavioral Research* (4th ed.). Mayfield Publishing, Mountian View, California.
- Craig, C. S., and McCann, J. M. (1978). Assessing Communication Effects on Energy Conservation. *Journal of Consumer Research* 5, 82-88.

- Cronbach, L. J. (1951). Coefficient Alpha and the Structure of Tests. *Psychometrika* 16, 297-334.
- Cronin, M. J. (1994). Doing Business on the Internet: How the Electronic Highway is Transforming American Companies. Van Nostrand Reinhold, New York, New York.
- Cross, R. (1994). Internet: The Missing Marketing Medium. Direct Marketing 57, 20-23.
- Czinkota, M. R., and Kotabe, M. (2001). *Marketing Management* (2nd ed.). South-Western Collge Publishing, Cincinnati, OH.
- Davis, F. D. (1989). Perceived Usefulness, Perceived Ease of Use, and User Acceptance of Informations Technology. *MIS Quarterly* 13, 319-340.
- Davis, F. D., Bagozzi, R. P., and Warshaaw, P. R. (1989). User Acceptance of Computer Technology: A comparison of Two Theoretical Models. *Management Science* 35, 982-1003.
- Davis, F. D. (1993). User Acceptance of Information Technology: System Characteristics, User Perceptions and Behavioral Impacts. *INternational Journal of Man-Machine Studies* 38, 475-487.
- Davis, F. D., Hohse, G. L., and Kottemann, J. E. (1994). Harmful Effects of Seemingly Helpful Information on Forecasts of Stock Earnings. *Journal of Economical Psychology* 15, 253-267.
- Day, E., Stafford, M. R., and Camacho, A. (1995). Research Note: Opportunities for Involvement Research: A Scale-Development Approach. *Journal of Advertising* 24, 69-75.
- Deighton, J., Romer, D., and McQueen, J. (1989). Using Drama to Persuade. *Journal of Consumer Research* 16, 335-343.
- Desharnais, P., Lu, J., and Radhakrishnan, T. (2002). Exploring Agent Support at the User Interface in Ecommerce Applications. *International Journal of Digital Library* 3, 284-290.
- Detlor, B., and Arsenault, C. (2002). Web Information Seeking and Retrieval in Digital Library Contexts: Towards an Intelligent Agent Solution. *Online Information Review* 26, 404-412.
- Dibb, S., Simkin, L., Pride, W. P., and Ferrell, O. C. (2001). Marketing Concepts and

Strategies (3rd ed.). Houghton-Mifflin Company, Boston, MA.

- Edmunds, A., and Morris, A. (2000). The Problem of Information Overload in Business Organizations: A Review on the Literature. *International Journal of Information Management* 20, 17-28.
- Einhorn, H., and Hogarth, R. (1978). Confidence in Judgment: Persistence of the Illusion of Validity. *Psychology Review* 85, 395-426.
- Emory, C. W., and Cooper, D. R. (1991). Business Research Methods (3rd ed.). Irwin Publishing, Homewood, IL.
- Feltham, T. S. (1994). Assessing Viewer Judgement of Advertisements and Vehicles: Scale Development and Validation. In Advances in Consumer Research (Roedder-John, C. T. A. a. D., ed, pp. 531-535. Association for Consumer Research, Provo, UT.
- Fishbein, M., and Ajzen, I. (1975). Belief, Attitude, Intention, and Behavior. Addison-Wesley, Reading, MA.
- Fishbein, M., Hennessy, M., Yzer, M., and Douglas, J. (2003). Can We Explain Why Some People Do and Some People Do Not Act on Their Intentions? *Psychology, Health & Medicine* 8, 3-19.
- Foner, L. N. (1993). What is an Agent, Anyway? A Sociological Case Study. *working* paper, MIT media labs
- Galletta, D. F., and Lederer, A. L. (1989). Some Cautions on the Measurement of User Information Satisfaction. *Decision Sciences* 20, 413-438.
- Gatain, A. W. (1994). Is User Satisfaction a Valid Measure of System Effectiveness? *Information & management* 26, 119-131.
- Gefen, D., Karahanna, E., and Straub, D. W. (2003). Inexperience and Experience with Online Stores: The Importance of TAM and Trust. *IEEE Transactions on Engineering Management* 50, 307-321.
- Ginzberg, M. J. (1981). Early Diagnosis of MIS Implementation Failure: Promising Results and Unanswered Questions. *Management Science* 27, 459-478.
- Gordon, M. E., Slade, L. A., and Schmitt, N. (1986). The "Science of Sophomore" Revisited: From Conjecture to Empiricism. Academy of Management Review 11, 191-207.

- Gorn, G. J. (1982). The Effects of Music in Advertising on Choice Behavior: A Classical Conditioning Approach. *Journal of Marketing* 46, 94-102.
- Gregor, S., and Benbasat, I. (1999). Explanations from Intelligent Systems: Theoretical Foundations and Implications for Practice. *MIS Quarterly* 23, 497-530.
- Grenci, R. T., and Todd, P. A. (2002). Solutions-Driven Marketing. *Communications of the ACM* 45, 64-71.
- Grise, M., and Guallupe, R. B. (1999/2000). Information Overload: Addressing the Productivity Paradox in Face-to-Face Electronic Meetings. *Journal of Management Information Systems* 16, 157-185.
- Grossman, R. P., and Till, B. D. (1998). The Persistence of Classically Conditioned Brand Attitude. *Journal of Advertising* 27, 23-31.
- Guttman, R. H., Moukas, A. G., and Maes, P. (1998a). Agent-Mediated Electronic Commerce: A Survey. *Knowledge Engrg. Review* 13, 147-160.
- Guttman, R. H., Mouksas, A. G., and Maes, P. (1998b). Agents as Mediators in Electronic Commerce. *Electronic Markets* 8, 22-27.
- GVU. (1999). *GVU's WWW User Survey*. Georgia Institute of Technology: Graphic, Visualization, and Usability Center.
- Hair, J. F., Jr., Anderson, R. E., Tatham, R. L., and Black, W. C. (1995). *Multivariate Data Analysis*. Prentice-Hall, Edgewood Cliffs, NJ.
- Hake, R. (1999). *E-commerce, A World of Opportunity*. The British Computer Society, Wiltshire,UK.
- Hanani, U., Shaira, B., and Shoval, P. (2001). Information Filtering: Overview of Issues, Research and Systems. *User Modeling and User-adapted Interaction* 11, 203-250.
- Harmon, H. H. (1976). *Modern Factor Analysis*. University of Chicago Press, Chicago, IL.
- Harrell, G. D., and Frazier, F. L. (1999). *Marketing, Connecting with Customers*. Prentice-Hall, Englewood Cliffs, NJ.
- Hartwick, J., and Barki, H. (1994). Explaining the Role of User Participation in Information System Use. *Management Science* 40, 440-465.

- Haubl, G., and Trifts, V. (2000). Consumer Decision Making in Online Shopping Environment: The Effects of Interactive Decision Aids. *Marketing Science* 19, 4-21.
- He, M. H., Jennings, N. R., and Leung, H.-F. (2003). On Agent-Mediated Electronic Commerce. *IEEE transactions on knowledge and data engineering* 15, 985-1003.
- Herbig, P. A., and Kramer, H. (1994). The Effect of Information Overload on the Innovation Chioce Process. *Journal of Consumer Marketing* 11, 45-54.
- Herlocker, J., Konstan, J. A., Borchers, A., and Riedl, J. (Year). "An Algorithmic Framework for Performing Collaborative Filtering." Paper presented at the SIGIR'99, Berkeley, CA, 1999.
- Herlocker, J. L., Konstan, J. A., and Riedl, J. (Year). "Explaining Collaborative Filtering Recommendations." Paper presented at the CSCW'00, Philadelphia, PA, Dec. 2-6, 2000.
- Hoffman, D. L., and Novark, T. P. (1996). Marketing in Hypermedia Computer-Mediated Environments: Conceptual Foundations. *Journal of Marketing* 60, 50-68.
- Hollnagel, E. (1987). Commentary: Issues in Knowledge-Based Decision Support. International Journal of Information Management 27, 743-751.
- Huang, Z., Chung, W. Y., Barbara, M., and Chen, H. C. (2004). A Graph Model of E-Commerce Recommender Systems. *Journal of the American Society for Information Science & Technology* 55, 259-275.
- Igbaria, M., and Machman, S. A. (1990). Correlates of User Satisfaction with End User Computing: An Exploratory Study. *Information & Management* 19, 73-82.
- Igbaria, M., Schiffman, S. J., and Wieckowski, T. J. (1994). The Respective Roles of Perceived Usefulness and Perceived Fun in the Acceptance of Microcomputer Technology. *Behavior & Information Technology* 13, 349-361.
- Iivari, J., and Ervasti, I. (1994a). User Information Satisfaction: IS Implementability and Effectiveness. *Information & Management* 27, 205-220.
- Iivari, J., and Ervasti, I. (1994b). User Information: IS Implementability and Effectiveness. *Information & management* 27, 205-220.
- Ives, B., Olson, M. H., and Baroudi, J. J. (1983). The Measurement of User Information

satisfaction. Communications of the ACM 26, 785-793.

- Jaccard, J. (1981). Toward Theories of Persuasion and Belief Change. Journal of Personality & Social Psychology 40, 260-269.
- Jacob, J. (1977). Information Load and Decision quality: Some Contested Issues. *Journal* of Marketing Research 14, 569-573.
- Jacob, J. (1984). Perspectives on Information Overload. *Journal of Consumer Research* 10, 432-436.
- Jain, S. P. (2003). Preference Consistency and Preference Strength: Processing and Judgmental Issues. *Journal of Applied Social Psychology* 33, 1088-1109.
- Jeffrey, M. B. (1997). Software Agents, Menlo Park, CA.
- Jiang, J. J., Klein, G., and Vedder, R. G. (2000). Persuasive Expert Systems: the Influence of Confidence and Discrepancy. *Computers in Human Behavior* 16, 99-109.
- Jobber, D. (2001). *Principles & Practice of Marketing*. McGraw-Hill International (HK) Limited, New York, NY.
- Johnson, M. A. (1989). Concern for Appropriateness Scale and Behavior Conformity. Journal of Personality Assessment 53, 567-574.
- Joines, J. L., Scherer, C. W., and Scheufele, D. A. (2003). Exploring Motivations for Consumer Web Use and Their Implications for E-commerce. *Journal of Consumer Marketing* 20, 90-108.
- Kahle, L. R., and Homer, P. M. (1985). Physical Attractiveness of the Celebrity Endorser: A Social Adaptation Perspective. *Journal of Consumer Research* 11, 954-961.
- Kahneman, D., and Tversky, A. (1974). Judgment Under Uncertainty: Heuristics and Biases. *Science* 185, 1124-1131.
- Karson, E. J., and Korgaonkar, P. K. (2001). An experimental investigation of Internet advertising and the Elaboration Likelihood Model. *Journal of Current Issues and Ressearch in Advertising* 23, 53-72.
- Kay, A. (1984). Computer Software. Scientific American 251, 53-59.

Keen, P., and Morton, M. S. (1987). Decision Support Systems: the Next Decade.

Decision Support Systems 3, 253-265.

- Keil, M., Beranek, P. M., and Konsynski, B. R. (1995). Usefulness and Ease of Use: Field Study Evidence Regarding Task Considerations. *Decision Support Systems* 13, 75-91.
- Keller, K. L., and Staelin, R. (1987). Effects of Quality and Quantity of Information on Decision Effectiveness. *Journal of Consumer Research* 14, 200-214.
- Keller, P. A., and Block, L. G. (1997). Vividness Effects: A Resource-Matching Perspective. *Journal of Consumer Research* 24, 295-304.
- Kerlinger, F. N. (1986). *Foundations of Behavioral Research* (3rd ed.). Harcourt Brace Jovanovich, New York, NY.
- Kleinmuntz, D. N., and Shchkade, D. A. (1993). Information Displays and Decision Processes. *Psychology Science* 4, 221-227.
- Komiak, S., and Benbasat, I. (Year). "Comparing Persuasiveness of Different Recommendation Agents as Customer Decision Support Systems in Electronic Commerce." Paper presented at the ITIP TC8/WG8.3 International Conference 2004, 2004.
- Kotler, P. (2003). *Marketing Management* (11th ed.). Prentice-Hall International Editions, Englewood Cliffs, NJ.
- Kottemann, J. E., and Remus, W. E. (1994). Computer-Assisted Decision Making: Performance, Beliefs and Ilusion of Control. *Organizational Behavior and Human Decision Processes* 57, 26-37.
- Koufaris, M., Kambil, A., and LaBarbera, P. A. (2001-2002). Consumer Behavior in Web-based Commerce: An Empirical Study. *International Journal of Electronic COmmerce* 6, 115-138.
- Kuehn, K. R., Khandekar, R. P., and Scott, C. R. (1996). The Effects of Marginality and Reward on Matrix Conflict. *Project Management Journal* 27,
- Laaksonen, P. (1994). Consumer Involvement. Routledge, London.
- Laczniak, R. N., Kempf, D. S., and Muehling, D. D. (1999). Advertising Message Involvement: the Role of Enduring and Situational Factors. *Journal of Current Issues and Ressearch in Advertising* 21, 51-61.

- Lee, P.-M. (2002). Behavioral Model of Online Purchasers in E-commerce Environment. *Electronice Commerce Research* 2, 75-85.
- Lee, S., M., and Kim, Y., R. (1995). An Empirical Study of the Relationships among End-User Information Systems Acceptance, Training, and Effectiveness. *Journal* of Management Information Systems 12, 189-202.
- Lee, W. P., Liu, C. H., and Lu, C. C. (2002). Intelligent Agent-Based Systems for Personalized Recommendations in Internet Commerce. *Expert systems with* applications 22, 275-285.
- Lennox, R. D., and Wolfe, R. N. (1984). Revision of the Self-Monitoring Scale. *Journal* of Peronality & Social Psychology 46, 1349-1364.
- Levy, S. (1993). Let Your Agents Do the Walking. MacWorld 42.
- Liang, T.-P., and Lai, H.-J. (2002). Effect of Store Design on Consumer Purchase: An Empirical Study of Online Bookstores. *Information & Management* 39, 431-444.
- Liao, Z., and Cheung, M. T. (2001). Internet-Based E-Shopping and Consumer Attitudes: An Empirical Study. *Information & Management* 38, 299-306.
- Liebermann, Y., and Stashevsky, S. (2002). Perceived Risks as Barriers to Internet and Ecommerce Usage. *Qualitative Market Research* 5, 291-300.
- Liu, C., Arnett, K. P., Capella, L. P., and Beatty, R. C. (1997). Web Sites of the Fortune 500: Facing Customers through Home Pages. *Information & management* 31, 335-345.
- Liu, C., and Arnett, K. P. (2000). Exploring the Factors Associated with Web Site Success in the Context of Electronic Commerce. *Information & Management* 38, 23-33.
- Locke, E. A. (1986). Generalizing from Laboratory to Field Settings. Lexington Books.
- Lowrey, T. M. (1998). The Effects of Syntactic Complexity on Adversiting Persuasiveness. *Journal of Consumer psychology* 7, 187-207.
- Lynch, J. G., and Ariely, D. (2000). Wine Online: Search Costs and Competition on Price, Quality, and Distribution. *Marketing Science* 19,
- Macias, W. (2003). A Beginning Look at the Effects of Interactivity, Product Involvement and Web Experience on Comprehension: Brand web sites as

Interactive Advertising. Journal of Current Issues and Ressearch in Advertising 25, 31-44.

- MacInnis, D. J., and Jaworski, B. J. (1989). Information Processing from Advertisments: Toward an Integrative Framework. *Journal of Marketing* 53, 1-23.
- MacInnis, D. J., and Park, C. W. (1991). The Differential Role of Characteristics of Music on High and Low Involvement Consumers' Processing of Ads. *Journal of Consumer Research* 18, 161-173.
- Maes, P., Butmann, R. H., and Moukas, A. G. (1999). Agents That Buy and Sell: Transforming Commerce as We Know It. *Communications of the ACM* 42, 81-91.
- Maheswaran, D., and Sternthal, B. (1990). The Effects of Knowledge, Motivation, and Type of Message on Ad Processing and Product Judgments. *Journal of Consumer Research* 17, 66-73.
- Malaviya, P., Kisielius, J., and Sternthal, B. (1996). The Effect of Type of Elaboration on Advertisment Processing and Judgment. *Journal of Marketing Research* 33, 410-421.
- Malhotra, N. K. (1982). Information Overload and Consumer Decision Making. *Journal* of Consumer Research 8, 419-431.
- Mao, J. Y., and Benbasat, I. (2000). The Use of Explanations in Knowledge-Based Systems: Cognitive Perspectives and a Process-Tracing Analysis. *Journal of Management Information Systems* 17, 153-179.
- McCabe, D. L. (1987). Buying Group Structure: Constriction at the Top. Journal of Marketing 51, 89-98.
- McGuire, W. J. (1969). The Nature of Attitude and Attitude Change. In *The Handbook of Psychology* (Aronson, G. L. a. E., ed, pp. 136-314. Addison-Wesley, Reading, MA.
- McGuire, W. J. (1985). Attitudes and Attitude Change. Random House, New York.
- McKnight, D. H., Choudhury, V., and Kacmar, C. (2002). The Impact of Initial Consumer Trust on Intentions to Transact with a Web Site: a Trust-Building Model. *The Journal of Strategic Information Systems* 11, 297-323.
- Meloy, M. G. (1997). The Role of Posiitve Affect in the Pre-decsional Distortion of

Product Information. Dissertation, Cornell University, US.

- Meyer, J. (1998). Informatino Overload in Marketing Management. *Marketing Intelligence & Planning* 16, 200-209.
- Miles, G. E., Howes, A., and Davis, A. (2000). A Framework for Understanding Human Factors in Web-based Electronic Commerce. *International Journal of Human-Computer Studies* 52, 131-163.
- Miller, H. E. (1966). Discussion of the Accounting Period Concept and Its Effect on Management Decisions. *Empirical Research in Accounting: Selected Studies, Journal of Accounting Research* 4, 15-17.
- Miller, J., and Doyle, B. A. (1987). Measuring Effectiveness of Computer Based Information Systems in the Finanical Service Sector. *MIS Quarterly* 11, 107-124.
- Miniard, P. W., Bhatla, S., Lord, K. R., Dickson, P. R., and Unnava, H. R. (1991). Picture-Based Persuasion Processes and the Moderating Role of Involvement. *Journal of Consumer Research* 18,
- Mitchell, A. A., and Olson, J. C. (1981). Are Product Attitude Beliefs the Only Mediator of Advertising Effects on Brand Attitude? *Journal of Marketing Research* 18, 318-332.
- Mitchell, A. A. (1986). The Effects of Verbal and Visual Components of Advertisements on Brand Attitudes and Attitudes Toward the Advertisement. *Journal of Consumer research* 13, 12-24.
- Montaner, M., Lopez, B., and de la Rosa, J. L. (2003). A Taxonomy of Recommender Agents on the Internet. *Artificial Intelligence Review* 19, 285-330.
- Morris, A. (1994). Online Company Database Selection: an Evaluation of Directories and CIDA (an Expert System). *Journal of Information Science* 20, 260-269.
- Mouksa, A., Zacharia, G., Guttman, R., and Maes, P. (2000). Agent-Mediated Electronic Commerce: A MIT Media Laboratory Perspective. *International Journal of Electronic Commerce* 4, 5-21.
- Moulin, B., Irandoust, H., Belanger, M., and Desbordes, G. (2002). Explanation and Argumentation Capabilities: Towards the Creation of More Persuasive Agents. *Artificial Intelligence Review* 17, 169-222.

Mowen, M., and Mowen, J. C. (1986). An Empirical Examination of the Biasing Effects

of Framing on Business Decisions. Decision Sciences 17,

Murch, R., and Johnson, T. (1999). *Intelligent Software Agents*. Prentice Hall, Upper Saddle River, NJ.

Negroponte, N. (1970). The Architecture Machine. MIT Press, Boston.

- Northcraft, G. B., and Neale, M. A. (1987). Experts, Amateurs, and Real Estate: An Anchoring-and-Adjustment Perspective on Property Pricing Decisions. *Organizational Behavior and Human Decision Processes* 39, 84-97.
- Nunnally, J. C. (1978). Psychometric Theory. McGraw-Hill, New York, NY.
- Nwana, H. S., and Azarmi, N. (1997). Software Agents and Soft Computing: toward Enhancing Machine Intelligence: Concepts and Applications. xiv, 298.
- O'Cass, A., and Fenech, T. (2003). Web Retailing Adoption: Exploring the Nature of Internet Users' Web Retailing Behavior. *Journal of Retailing and Consumer Services* 10, 81-94.
- O'Keefe, R. M., and McEachern, T. (1998). Web-Based Customer Decision Support Systems. *Communications of the ACM* 41, 71-78.
- Olney, T. J., Holbrook, M. B., and Batra, R. (1991). Consumer Responses to Advertising: The Effects of Ad Content, Emotions, and Attitude Toward the Ad on Viewing Time. *Journal of Consumer Research* 17, 440-453.
- Ozok, A. A. (1997). Measuring Consistency of Home Page Design and Its Effects on Performance and Satisfaction, School of Industrial Engineering, Purdue University, W. Lafayette, IN.
- Ozok, A. A. (2000). Measuring Consistency of Webpage Design and Its Effects on Performance and Satisfaction. *Ergonomics* 43, 443-460.
- Pare, G., and Elam, J. J. (1995). Discretionary Use of Personal Computers by Knowledge Workers: Tesing of a Social Psychology Theoretical Model. *Behavior & Information Technology* 14, 215-228.
- Payne, J., Bettman, J., and Johnson, E. (1993). *The Adaptive Decision Maker*. Cambridge University Press, New York.
- Peter, J. P. (1979). Relaibility: A Review of Psychometric Basics and Recent Marketing Practices. *Journal of Marketing Research* 16, 6-17.

- Peter, J. P. (1981). Construct Validity: A Review of Basic Issues and Marketing Practices. *Journal of Marketing Research* 18, 133-145.
- Petty, R. E., and Cacioppo, J. T. (1981). Issue Involvement as a Moderator of the Effects on Advertising Content and Context. In *Advances in Consumer Research* (M., K. B., ed, pp. 20-24. Association for Consumer Research, Ann Arbor, MI.
- Petty, R. E., Shumann, D., and Cacioppo, J. T. (1983). Central and Peripheral Routes to Advertising Effectiveness: The Moderating Role of Involvement. *Journal of COnsumer Research* 10, 135-146.
- Petty, R. E., and Cacioppo, J. T. (1984). The Effects of Involvement on Responses to Argument Quantity and Quality: Central and Peripheral Routes to Persuasion. *Journal of Personality & Social Psychology* 46, 69-81.
- Petty, R. E., and Cacioppo, J. T. (1986). The Elaboration Likelihood Model of Persuasion, in L. Berkowitz (ed.). *Advances in Experimental Social Psychology* 19, 123-205.
- Petty, R. E., Kasmer, J. A., Haugtvedt, C. P., and Cacioppo, J. T. (1987). Source and Message Factors in Persuasion: a Reply to Stiff's Critique of the Elaboration Likelihood Model. *Communications Monogrpaghs* 57, 233-249.
- Pornpitakpan, C. (2004). The Persuasiveness of Source Credibility: A Critical View of Five Decades. *Journal of Applied Social Psychology* 34, 243-282.
- Powell, R. (2001). DM Review: A 10 Year Journey. DM Review
- Power, D. J. (Year). "Web-Based and Model-Driven Decision Support Systems: Concepts and Issues." Paper presented at the Americas Conference on Information Systems 2000, Long Beach, California, August 10th-13th, 2000.
- Pyle, R. (1996). Electronic Commerce and the Internet. *Communications of the ACM* 39, 22-23.
- Quester, P. G., and Smart, J. (1998). The Influence of Consumption Situation and Product Involvement over Consumers' use of product Attribute. *Journal of Consumer Marketing* 15, 220-238.
- Rashid, A. M., Albert, I., Cosley, D., Lam, S. K., McNee, S., Konstan, J. A., and Riedl, J. (Year). "Getting to Know You: Learning New User Preferences in Recommendation Systems." Paper presented at the International Conference on

Intelligent User Interfaces 2002, San Francisco, CA, 2002.

Reardon, K. K. (1981). Persuasion: Theory and context. Sage, Beverly Hills, CA.

- Reardon, K. K. (1987). Interpersonal Communication: Where Minds Meet. Wadsworth, Belmont, CA.
- Reardon, K. K. (1991). Persuasion in Practice. Sage Publications.
- Reibstein, D. J. (2002). What Attracts Customers to Online Stores, and What Keeps Them Coming Back? *Journal of the Academy of Marketing Science* 30, 465-473.
- Reichheld, F., and Sasser, W. E. (1990). Zero Defection: Quality Comes to Services. *Harvard Business School Review* 5, 105-111.
- Reichheld, F. (1993). Loyalty-Based Management. Harvard Business School Review 2, 64-73.
- Resnick, P., Iacovou, N., Suchak, M., Bergstrom, P., and Riedl, J. (Year). "Grouplens: An Open Architecture for Collaborative Filtering of Netnews." Paper presented at the ACM CSCW'94 Conference on Computer-Supported Cooperative Work, New York, New York, 1994.
- Robinson, P. B., C., H. J., and Hunt, H. K. (1991). Entrepreneurial Research on Student Subjects Does Not Generalize to Real World Entrepreneurs. *Journal of Small Business Management* 29,
- Rudowsky, I. S. (2004). Intelligent Agent. Communications of the AIS 14, 275-290.
- Salton, G., and Buckley, C. (1988). Term Weighting Approaches in Automatic Text Retrieval. *Information Processing and Management* 24, 513-523.
- Sarwar, B. M., Konstan, J. A., Borchers, A., Herlocker, J., Miller, B., and Riedl, J. (Year). "Using Filtering Agents to Improve Prediction Quality in the GroupLens Research Collaborative Filerting System." Paper presented at the ACM 1998 Conference on Computer Supported Collaborative Work, New York, 1998.
- Schafer, J. B., Konstan, J., and Riedl, J. (Year). "Recommender Systems in E-Commerce." Paper presented at the ACM Conference on Electronic Commerce 1999, 1999a.
- Schafer, J. B., Konstan, J. A., and Riedl, J. (Year). "E-commerce Recommendation Applications." Paper presented at the ACM Conference on Electronic Commerce

1999, 1999b.

- Sears, D. O. (1986). College Sophomores in the Laboratory: Influences of a Narrow Data Base on Social Psychology's View of Human Nature. *Journal of Peronality & Social Psychology* 51, 515-530.
- Sewall, M. A., and Sarel, D. (1986). Characteristics of Radio Commercials and Their Recall Effectiveness. *Journal of Marketing* 50, 52-60.
- Shardanand, U., and Maes, P. (Year). "Soical Information Filtering: Algorithms for Automating 'Word of Mouth'." Paper presented at the ACM CHI '95 Conference on Human Factors in Computing Systems, New York, 1995.
- Shim, J. P., Warkentin, M., Courtney, F. J., Power, D. J., Sharda, R., and Carlsson, C. (2002). Past, Present, and Future of Decision Support Technology. *Decision Support Systems* 33, 111-126.
- Shugan, S. M. (1980). The Cost of Thinking. Journal of Consumer Research 7, 99-111.
- Sinha, R., and Swearingen, K. (2001). Comparing Recommendations Made by Online Systems and Friends. Delos-NSF workshop on "Personalization and Recommender Systems in Digital Libraries".
- Sinha, R., and Swearingen, K. (Year). "The Role of Transparency in Recommender Systems." Paper presented at the Conference on Human Factors in Computing Systems Archive, Minneapolis, Minnesota, USA, 2002.
- Solomon, M. R., and Stuart, E. W. (2003). *Marketing, Real People, Real Choices* (3rd ed.). Prentice-Hall, Englewood Cliffs, NJ.
- Stafford, M. R. (1998). Advertising, Sex-Typed Services: The Effects of Sex, Service Type, and Employee Type on Consumer Attitudes. *Journal of Advertising* 27,
- Stayman, D. M. (1990). The Role of Mood in Advertising Effectiveness. Journal of Consumer Research 17,
- Sternthal, B., Dholakia, R., and Leavitt, C. (1978). The Persuasive Effect f Source Credibility: Tests of Cognitive Response. *Journal Of Consumer Research* 4, 252-260.
- Stewart, D. W. (1999). Advertising Wearout: What and How You Measure It Matters. Journal of Advertising Research 39, 39-42.

- Strahan, E. J., Spencer, S. J., and Zanna, M. P. (2002). Subliminal Priming and Persuasion: Striking While the Iron is Hot. *Journal of Experimental Social Psychology* 38, 556-569.
- Straub, D. W. (1989). Validating Instruments in MIS Research. *MIS Quarterly* 13, 147-169.
- Strutton, D., and Lumpkin, J. R. (1994). Problem- and Emotion-Focused Coping Dimensions and Sales Presentation Effectiveness. *Journal of The Academy of Marketing Science* 22, 28-37.
- Suh, B., and Han, I. (2002). Effect of Trust on Customer Acceptance of Internet Banking. *Electronic Commerce Research and Applications* 1, 247-263.
- Swait, J., and Erdem, T. (2002). The Effects of Temporal Consistency of Sales Promotions and Availability on Consumer Choice Behavior. *Journal of Marketing Research* 39,
- Swaminathan, V. (2003). The Impact of Recommendation Agents on Consumer Evaluation and Choice: The Moderating Role of Category Risk, Product Complexity, and Consumer Knowledge. *Journal of Consumer Psychology* 13, 93-101.
- Tabachnick, B. G., and Fidell, L. S. (1983). Using Multivariate Statistics. New York Harper & Row.
- Tetzeli, R. (1994). Electronic Storefronts on the Internet. Fortune 130, 191.
- Todd, P., and Benbasat, I. (1992). The Use of Information in Decision Making: An Experimental Investigation of the Impact of Computer-Based Decision Aids. *Management Information Systems Quarterly* 16, 373-393.
- Torson, E., and Page, T. (1988). Effects of Product Involvement and Emotional Commercials on Consumers' Recall and Attitude. In *Nonverbal Communication in Advertising* (Steward, S. H. a. D. W., ed, pp. 111-126. Lexington Books, Lexington, MA.
- Turban, E., Lee, J., King, D., and Chung, H. M. (2004). *Electronic Commerce: A Managerial Perspective*. Prentice Hall.
- Tversky, A., and Kahneman, D. (1974). Judgment Under Uncertainty. *Science* 185, 1124-1130.

- Walizer, M. H., and Wienir, P. L. (1978). Research Methods and Analysis: Searching for Relationships. Harpers-Row, New York, NY.
- Wang, W., and Benbasat, I. (2003). Impact of Explanation on Trust in and Intention to Adopt Online Recommendation Agents.
- Wang, W., and Benbasat, I. (Year). "Trust and TAM for Online Recommendation Agents." Paper presented at the Tenth Americas Conference on Information Systems, New York, New York, 2004.
- Wang, W., and Benbasat, I. (2005). Turst in and Adoption of Online Recommendation Agents. *Journal of the Association for Information Systems* 6, 72-101.
- Waters, L. K., and Collins, M. (1984). Effect of Pricing Conditions on Reference Reversals by Business Students and Managers. *Journal of Applied Psychology* 69,
- West, P. M., Ariely, D., Bellman, S., Bradlow, E., Huber, J., Johnson, E., Kahn, B., Little, J., and Schkade, D. (1999). Agents to the Rescue? *Marketing Letters* 10, 285-300.
- Will, R. P. (1992). Individual Differences in the Performance and Use of an Expert System. *International Journal of Man-Machine Studies* 37, 173-190.
- Wolf, J., Aggarwal, C., Wu, K.-L., and Yu, P. (Year). "Horting Hatches an Egg: A New Graph-Theoretic Approach to Collaborative Filtering." Paper presented at the ACM SIGKDD International Conference on Knowledge Discovery & Data Mining, San Diego, CA, 1999.
- Wooldridge, M., and Jennings, N. R. (1995). Intelligent Agents: Theory and Practice. *The Knowledge Engineering Review* 10, 115-152.
- Yalch, R. F., and Yalch, R. E. (1984). The Effect of Numbers on the Route to Persuasion. *Journal of Consumer Research* 11, 522-527.
- Yavas, U. (1994). Resaerch Notes: Students as Subjects in Advertising and Marketing Research. *iNternational Marketing Review* 11,
- Ye, L. R., and Johnson, P. E. (1995). The Impact of Explanation Facilities on User Acceptance of Expert-Systems Advice. *MIS Quarterly* 19, 157-172.
- Yoo, C. Y., and Stout, P. A. (Year). "Factors Affecting Users' Interactivity with the Web Site and the Consequences of Users' Interactivity." Paper presented at the the 2001 Conference of the American Academy of Advertising, Villanova, PA, 2001.

- Yoon, S. N., and Lee, Z. (Year). "The Impact of the Web-Based Product Recommendation Systems from Previous Buyers on Consumers' Purchasing Behavior." Paper presented at the The tenth Americas Conference on Information Systems, New York, New York, August, 2004.
- Yuthas, K., and Young, S. (1998). Material Matters: Assessing the Effectiveness of Materials Management Information System. *Information & management* 33, 115-124.
- Yzer, M. C., Hennessy, M., and Fishbein, M. (2004). The Usefulness of Perceived Difficulty for Health Research. *Psychology, Health & Medicine* 9, 149-163.
- Zaichowsky, J. L. (1985). Measuring the Involvement Construct. Journal of Consumer Research 12, 341-352.
- Zinkhan, G. M., and Leigh, J. H. (1986). Dimensinal Relations of Aided Recall and Recognition. *Journal of Advertising* 15, 38-46.
- Zinkhan, G. M., and Locander, W. B. (1988). ESSCA: A Multi-Dimensional Analysis Tool for Marketing Research. *Journal of the Academy of Marketing Science* 16, 36-46.
- Zwass, V. (1996). Electronic Commerce: Structures and Issues. *International Journal of Electronic Commerce* 1, 3-23.
APPENDIX

A. RECOMMENDER SYSTEM EXPERIMENT SURVEY

INSTRUCTIONS

Thank you for participating in this experiment. Your answers are important for the study. Please be candid and cooperative in the experiment and return the completed survey when you leave the classroom. Don't discuss about the experiment during or after the session. Sign on both page of the Informed Consent form and keep one copy for your own reference.

In the entire computer experiment, you are expected to perform as quickly and as accurately as possible. A 5 minute training session will be conducted before the computer experiment. Definitions frequently used in the questionnaire are provided on the next page.

The experiment consists of four different tasks, and instructions for each task are presented before the start of each task. You should follow the instructions for each task. Please do not open or close any windows for purposes not related to the experiment.

In all of the surveys you are to complete today, we ask questions that make use of rating scales with 7 places. Please <u>circle a number</u> which represents your perception of the recommender system you just used. For example, if you believe that WebCT is a good learning tool and would like to take another course using WebCT in the future, you might respond to the statement below by circling the corresponding number.

		Strong Disagr	ly ee				Strongly Agree			
1.	I would like to enroll in a course using WebCT next semester.	1	2	3	4	5 (6) 7		

You can now start the experiment.

DEFINITIONS

- 1) **Recommender systems** are the interactive decision support systems that assist you in the initial screening of alternatives available in online stores and provide recommendations for product search and selection.
- 2) Appropriateness refers to your perception of the fitness and desirability of recommendations given by the recommender system.
- **3)** Consistency refers to whether the recommendations from a recommendation system are consistent with your existing knowledge.
- 4) Effectiveness refers to the perceived degree of the recommender system's ability to clearly understand and address your need and concerns.
- **5) Persuasiveness** refers to the extent to which you are moved or influenced by the recommender system's reasoning to a belief that the recommended products best fit your personal needs.

TASK 1

- 1. Open the browser Internet Explorer;
- 2. Visit http://www.activebuyersguide.com by typing in the website address;
- 3. Search for a Notebook computer you plan on purchasing soon by going to the Notebook computer category;
- 4. Choose the product attributes you think is the most important to you from the product attributes given;
- 5. Click recommend;
- 6. Viewing the recommended product which is marked as "best match";
- 7. If the system cannot find a product for your preference, adjust the preference criteria and search again;
- 8. Complete the questions on the next pages.

		Stror Disa	ngly gree				Stro Ag	ongly gree
1.	The product recommendation is consistent with your guidance.	1	2	3	4	5	6	7
2.	The recommendation output is consistent with entry requirement.	1	2	3	4	5	6	7
3.	The information in the recommend-ation appears to be consistent with human stereotypes.	1	2	3	4	5	6	7
4.	Where rating, symbols, product categories etc. are used, they are consistent with your ability to understand.	1	2	3	4	5	6	7
5.	The wording of the recommend- ation is familiar to you.	1	2	3	4	5	6	7
Ef	fectiveness							
1.	The recommender system was able to convince me that it understands my unique need and concerns.	1	2	3	4	5	6	7
2.	The recommender system was able to work out recommendations to my questions or objections.	1	2	3	4	5	6	7
3.	The recommender system communicates the product recommendations clearly and concisely.	1	2	3	4	5	6	7
4.	The recommender system makes effective use of aids (charts, pictures, audio-visuals) to improve the recommendation.	1	2	3	4	5	6	7

5.	The recommendation makes me	1	2	3	4	5	6	7	
	want to buy things from this site again.								

137

6.	The recommender system	Strongly	Strongly
	demonstrates identification and	Disagree	Agree
	understanding of my real need and	1 2 3 4 5 6	7
	concerns.		

Appropriateness

Over all the recommendations given by the recommender systems are:

1.	rational	1	2	3	4	5	6	7
2.	informative	1	2	3	4	5	6	7
3.	deal with facts	1	2	3	4	5	6	7
4.	knowledgeable	1	2	3	4	5	6	7
5.	logical	1	2	3	4	5	6	7

Persuasiveness

1.	believable	1	2	3	4	5	6	7
2.	credible	1	2	3	4	5	6	7
3.	trustworthy	1	2	3	4	5	6	7
4.	reliable	1	2	3	4	5	6	7

5.	dependable	1	2	3	4	5	6	7
6.	affects my feelings	1	2	3	4	5	6	7
7.	touch me emotionally	1	2	3	4	5	6	7
8.	stimulating	1	2	3	4	5	6	7
9.	reach out to me	1	2	3	4	5	6	7
10.	stirring	1	2	3	4	5	6	7
11.	moving	1	2	3	4	5	6	7
12.	exciting	1	2	3	4	5	6	7

Task 2

- 1. Open the browser Internet Explorer;
- 2. Visit <u>http://www.amazon.com</u> by typing the website address given;
- 3. Go to "**personalized recommendation**" and register **as a new user** by entering your email address. Use a different email address if you have registered at amazon.com before;
- 4. Enter you name and password;
- 5. Click on "start recommendation wizard";
- 6. Check the store category "Computers" and continue;
- 7. Type in a computer type <u>**OR**</u> a trusted brand you plan on purchasing soon, and continue;
- 8. Go through **all relevant items** provided and check detail specifications and explanations if needed. Don't rate any item before checking the details because the saved rating will disappear!
- 9. Rate **relevant items** given by clicking on the star-scale for each item. Click on "**rate more items**" if you think it is necessary;
- 10. Click finish to see your recommendations;
- 11. View the product that matches your choice by checking the product specification from the recommendation list;
- 12. Complete the questions on the next pages.

1.	The product recommendation is	Stror Disa	ngly gree				Str As	ongly gree
	consistent with your guidance.	1	2	3	4	5	6	7
2.	The recommendation output is consistent with entry requirement.	1	2	3	4	5	6	7
3.	The information in the recommendation appears to be consistent with human stereotypes.	1	2	3	4	5	6	7
4.	Where rating, symbols, product categories etc. are used, they are consistent with your ability to understand.	1	2	3	4	5	6	7
5.	The wording of the recommend- ation is familiar to you.	1	2	3	4	5	6	7
<u>Effect</u>	iveness							
1.	The recommender system was able to convince me that it understands my unique need and concerns.	1	2	3	4	5	6	7
2.	The recommender system was able to work out recommendations to my questions or objections.	1	2	3	4	5	6	7
3.	The recommender system communicates the product recommendations clearly and concisely.	1	2	3	4	5	6	7

									-
4.	The recommender system makes effective use of aids (charts, pictures, audio-visuals)	1	2	3	4	5	6	7	
	to improve the								
	recommendation.								
5.	The recommendation makes me want to buy things from this site again.	1	2	3	4	5	6	7	
	C C	_	_				_		

6. The recommender system Disagree Agree demonstrates identification and 1 2 3 4 5 6 7 understanding of my real need and concerns.

<u>Appropriateness</u>

Over all the recommendations given by the recommender systems are:

1.	rational	1	2	3	4	5	6	7
2.	informative	1	2	3	4	5	6	7
3.	deal with facts	1	2	3	4	5	6	7
4.	knowledgeable	1	2	3	4	5	6	7
5.	logical	1	2	3	4	5	6	7

Persuasiveness

1.	believable	1	2	3	4	5	6	7
2.	credible	1	2	3	4	5	6	7
3.	trustworthy	1	2	3	4	5	6	7
4.	reliable	1	2	3	4	5	6	7

5.	dependable	1	2	3	4	5	6	7
6.	affects my feelings	1	2	3	4	5	6	7
7.	touch me emotionally	1	2	3	4	5	6	7
8.	stimulating	1	2	3	4	5	6	7
9.	reach out to me	1	2	3	4	5	6	7
10.	stirring	1	2	3	4	5	6	7
11.	moving	1	2	3	4	5	6	7
12.	exciting	1	2	3	4	5	6	7

Task 3

- 1. Open the browser Internet Explorer;
- 2. Visit <u>http://www.amazon.com</u> by typing the website address given;
- 3. Go to "**personalized recommendation**" and register **as a new user** by entering your email address. Use a different email address if you have registered at amazon.com before;
- 4. Enter you name and password;
- 5. Click on "start recommendation wizard";
- 6. Check the store category "Electronics" and continue;
- 7. Type in DVD player <u>OR</u> a trusted brand of electronics you plan on purchasing soon, and continue;
- 8. Go through **all** items provided and check detail specifications and explanations if needed. Don't rate any item before checking the details because the saved rating will disappear!
- 9. Rate all items given by clicking on the star-scale for each item. Click on "rate more items" if you like;
- 10. Click finish to see your recommendations;
- 11. View the product that matches your choice by checking the product specification from the recommendation list;
- 12. Complete the questions on the next pages.

1.	The product recommendation is	Stror Disa	ngly gree				Str As	ongly gree
	consistent with your guidance.	1	2	3	4	5	6	7
2.	The recommendation output is consistent with entry requirement.	1	2	3	4	5	6	7
3.	The information in the recommendation appears to be consistent with human stereotypes.	1	2	3	4	5	6	7
4.	Where rating, symbols, product categories etc. are used, they are consistent with your ability to understand.	1	2	3	4	5	6	7
5.	The wording of the recommend- ation is familiar to you.	1	2	3	4	5	6	7
<u>Effect</u>	iveness							
1.	The recommender system was able to convince me that it understands my unique need and concerns.	1	2	3	4	5	6	7
2.	The recommender system was able to work out recommendations to my questions or objections.	1	2	3	4	5	6	7
3.	The recommender system communicates the product recommendations clearly and concisely.	1	2	3	4	5	6	7

									145
4.	The recommender system makes effective use of aids (charts, pictures, audio-visuals) to improve the recommendation.	1	2	3	4	5	6	7	
		Stron Disag	igly gree					Strongly Agree	
5.	The recommendation makes me want to buy things from this site again.	1	2	3	4	5	6	7	
6.	The recommender system demonstrates identification and understanding of my real need and concerns.	1	2	3	4	5	6	7	

Appropriateness

Over all the recommendations given by the recommender systems are:

1.	rational	1	2	3	4	5	6	7
2.	informative	1	2	3	4	5	6	7
3.	deal with facts	1	2	3	4	5	6	7
4.	knowledgeable	1	2	3	4	5	6	7
5.	logical	1	2	3	4	5	6	7

Persuasiveness

1.	believable	1	2	3	4	5	6	7
2.	credible	1	2	3	4	5	6	7
3.	trustworthy	1	2	3	4	5	6	7
4.	reliable	1	2	3	4	5	6	7

5.	dependable	1	2	3	4	5	6	7
6.	affects my feelings	1	2	3	4	5	6	7
7.	touch me emotionally	1	2	3	4	5	6	7
8.	stimulating	1	2	3	4	5	6	7
9.	reach out to me	1	2	3	4	5	6	7
10.	stirring	1	2	3	4	5	6	7
11.	moving	1	2	3	4	5	6	7
12.	exciting	1	2	3	4	5	6	7

Task 4

- 1. Open the browser Internet Explorer;
- 2. Visit <u>http://www.activebuyersguide.com</u> by typing in the website address;
- 3. Search for a **DVD** player you plan on purchasing soon by going to the **DVD** player category;
- 4. Choose the product attributes you think is the most important to you for the product attributes given;
- 5. Click recommend;
- 6. Viewing the recommended product which is marked as "best match";
- 7. If the system cannot find a product for your preference, adjust the preference criteria and search again;
- 8. Complete the questions on the next pages.

1.	The product recommendation is	Stror Disa	ngly gree				Str As	ongly gree
	consistent with your guidance.	1	2	3	4	5	6	7
2.	The recommendation output is consistent with entry requirement.	1	2	3	4	5	6	7
3.	The information in the recommendation appears to be consistent with human stereotypes.	1	2	3	4	5	6	7
4.	Where rating, symbols, product categories etc. are used, they are consistent with your ability to understand.	1	2	3	4	5	6	7
5.	The wording of the recommend- ation is familiar to you.	1	2	3	4	5	6	7
<u>Effect</u>	iveness							
1.	The recommender system was able to convince me that it understands my unique need and concerns.	1	2	3	4	5	6	7
2.	The recommender system was able to work out recommendations to my questions or objections.	1	2	3	4	5	6	7
3.	The recommender system communicates the product recommendations clearly and concisely.	1	2	3	4	5	6	7

									149
4.	The recommender system makes effective use of aids (charts, pictures, audio-visuals) to improve the recommendation.	1	2	3	4	5	6	7	
5.	The recommendation makes me want to buy things from this site again.	Stron Disaş 1	ngly gree 2	3	4	5	6	Strongly Agree 7	
6.	The recommender system demonstrates identification and understanding of my real need and concerns	1	2	3	4	5	6	7	

Appropriateness

Over all the recommendations given by the recommender systems are:

1.	rational	1	2	3	4	5	6	7
2.	informative	1	2	3	4	5	6	7
3.	deal with facts	1	2	3	4	5	6	7
4.	knowledgeable	1	2	3	4	5	6	7
5.	logical	1	2	3	4	5	6	7

Persuasiveness

1.	believable	1	2	3	4	5	6	7
2.	credible	1	2	3	4	5	6	7
3.	trustworthy	1	2	3	4	5	6	7
4.	reliable	1	2	3	4	5	6	7

5. dependable	1	2	3	4	5	6	7
6. affects my feelings	1	2	3	4	5	6	7
7. touch me emotionally	1	2	3	4	5	6	7
8. stimulating	1	2	3	4	5	6	7
9. reach out to me	1	2	3	4	5	6	7
10. stirring	1	2	3	4	5	6	7
11. moving	1	2	3	4	5	6	7
12. exciting <u>Product Involvement</u>	1	2	3	4	5	6	7

Using the following scale, please indicate how much you:

No	book Computer very little					vei	y much	
1.	use Notebook computer	1	2	3	4	5	6	7
2.	are involved with Notebook Computer	1	2	3	4	5	6	7
3.	are a Notebook computer expert	1	2	3	4	5	6	7
4.	are interested in Notebook computer, relative to other people	1	2	3	4	5	6	7
D	VD Player							
1.	use DVD player	1	2	3	4	5	6	7
2.	are involved with DVD player	1	2	3	4	5	6	7
3.	are a DVD player expert	1	2	3	4	5	6	7
4.	are interested in DVD player, relative to other people	1	2	3	4	5	6	7

Demographics

- 1. What is your age? _____ years old.
- 2. What is your gender? (Please circle one)

Male Female

3. What is your program? (Please circle one)

	Freshman	Sophomore	Junior	Senior	Graduate
4.	What is your ethnic	city? (Please cir	cle one)		

White/Caucasian African American Asian/Asian Islander Other

5. Which one best describe your computer usage experience? (Please circle one)

<1 year 1- <5 years 5-10 years >10 years

6.	I have purchased	Strongly Strong Disagree Agree						congly ree
	products/services through the Internet very often.	1	2	3	4	5	6	7
7.	I have used the recommender systems very often when buying things online.	1	2	3	4	5	6	7

8. What do you think the purpose of this experiment is?

Thank you for your participation!